# Long Term Effects of Wildfire on Permafrost Stability and Carbon Cycling in Northern Peatlands

by

Carolyn Gibson

A thesis submitted in partial fulfillment of the requirements for the degree of

Master of Science

in

Forest Biology and Management

Department of Renewable Resources

University of Alberta

© Carolyn Gibson, 2017

#### Abstract

Changing fire dynamics and increasing global temperatures are causing changes to the fire regime and permafrost stability in the Arctic. Models have separately predicted the widespread thawing of permafrost and increasing magnitude and intensity of wildfires over the next century. However, while it is evident that wildfire and permafrost are both dominant controls on carbon dynamics in the boreal, less is known about the potential effects of wildfire to cause increased permafrost thaw and to affect soil carbon stocks. To assess the role of wildfire in future permafrost stability and carbon storage in permafrost peatlands of northwestern Canada, I addressed the following questions; 1) to what extent does fire accelerate permafrost thaw in discontinuous permafrost regions and 2) what is the magnitude of carbon released from depth as a result of fire-induced permafrost thaw on peat plateaus. This research was conducted at a variety of sites located within the Northwest Territories that burned 2,3,9,14,15,21,33,40 and 48 years ago, and six, nearby unburned sites. Field data was complemented with the use of remotely sensed data to determine the extent of fire-induced permafrost thaw. Soil respiration was measured on a subset of these sites. The results of these studies, summarized in this thesis, find that wildfire destabilizes the post-fire soil thermal regime of peat plateaus manifested by deeper active layers and widespread formation of taliks, persisting for up to 30 with the most pronounced effect being 10-20 following fire. This also appears to result in ~3 times the rate of recent thermokarst formation, making wildfire responsible for ~25% of permafrost thaw in the past 30 years. As a result of fire induced deepening of the active layer and increased soil temperatures, soil respiration at depth was stimulated, representing a nearly four times greater respiration of old compared to unburned sites. Surprisingly though, I find that this is not enough to offset the importance of surface labile carbon pools, to soil respiration, that are burned off during the fire event, despite the mobilization and release of deep, old, previously stored carbon. This thesis concludes by highlighting the importance of wildfire as a driver of permafrost stability and old carbon storage capabilities and emphasizes the importance of it in global permafrost and carbon cycling models.

#### Preface

This thesis is an original work of Carolyn Gibson. Chapter 1 provides an introduction to permafrost dynamics and carbon cycling in the Taiga plains and an overview of each thesis chapter. Chapter 2 has been formatted for submission to *Proceedings in the National Academy of Science*, and is currently *in preparation*. Chapter 3 has been formatted for *Journal of Geophysical Research: Biogeoscience* and is currently *in preparation*. Finally, Chapter 4 summarizes and synthesizes the finding from this work.

Co-authors made important contributions to each dissertation chapter. For chapter 2, L. Chasmer assisted with remote sensing sampling design and classification technical assistance, D. Thompson with the design and technical advice on measuring ground heat fluxes, and B. Quinton and M. Flannigan with concept formation and manuscript composition. D. Olefeldt was the supervisory author and assisted at all parts. For chapter 3, C. Estop Aragones assisted with sampling design and M. Flannigan and D. Olefeldt were supervisory authors.

Funding and support for this research was provided by the National Science and Engineering Research Council Discovery grant (RGPIN-2016-04688), the Campus Alberta Innovates Program, the University of Alberta Northern Research Awards, and the Northern Scientific Training Program.

# **Acknowledgements**

I would like to start by acknowledging this research was completed in the Decho Region and thank the Decho First Nations whose histories, languages, and culture continue to influence our diverse research community.

This body of work is a culmination of the passion and efforts of a great number of people, without whom I would have never completed this thesis. I would like to thank the following people for not only their help, but for making my life as a graduate student more than just a thesis. For making it an experience that has shaped me into a better researcher, a better learner, a better listener, and most importantly, a better person.

I would like to thank my advisors Dr. David Olefeldt and Dr. Mike Flannigan for their guidance, support, and encouragement throughout my degree. I am also grateful to my thesis committee member, Dr. Daniel Thompson whose insightful comments and suggestions improved my experimental design and end written thesis.

To my lab mates and field assistants, Liam Heffernan, McKenzie Kuhn, Kathryn Burd, Cristian Estop Aragones, Jessi Steinke, and Michael Barbeau, I thank you as I would never have been able to stab the ground 10,000 times if it weren't for you. Some of my fondest memories will be from summers I spent with you learning the fire is hot, it melts ice. To François-Nicolas Robinne I thank you for all your help and expertise with ArcGIS.

I would like to thank those funding agencies without whom this work would not have been possible. Natural Science and Engineering Research Council grant (RGPIN-2016-04688), the Campus Alberta Innovation Program, the University of Alberta Northern Research Awards, and the Northern Scientific Training Program all supported this research. A special thank you to UAlberta North for creating endless opportunities and a hub for northern scientists.

I would also like to extend my most sincere gratitude to my friends, my family, and loved ones. Without your constant support and encouragement, this research would not have been possible. You have kept me balanced and have helped me keep in focus those things in life that are most important.

# **Table of Contents**

Chapter 1: Introduction	1
1.1 Overview of permafrost peatlands	1
1.2 Permafrost and the development of peat complexes.	1
1.3 Wildfire and the Boreal Forest.	3
1.4 Interaction between wildfire and permafrost	4
1.5 Carbon cycling in the Boreal.	5
1.6 Research Objectives	7
1.7 Literature Cited	9
Chapter 2: Wildfire as a major driver of recent permafrost thaw in boreal peatlands.	16
2.1 Abstract	16
2 2 Significance statement	16
2.3 Introduction	
2.4 Results and discussion	18
2.4.1 Effects of wildfire on peat plateau soil thermal regimes	18
2.4.2 Effects of wildfire on young thermokarst bog expansion	21
2.5 Conclusion	23
2.6 Materials and Methods	24
2.6.1 Site selection for soil thermal regime monitoring	24
2.6.2 Depth to frost table monitoring	24
2.6.3 Air and soil temperature monitoring	25
2.6.5 Image acquisition and preparation for supervised peatland classification	25
2.6.6 Supervised peatland classification	26
2.6.7 Accuracy assessment of peatland classification	26
2.6.8 Estimating trajectories of young thermokarst development	26
2.6.9 Spatial scaling of peat plateau loss due to thermokarst development	26
2.7 Acknowledgements	27
2.8 References.	27
2.9 SI Appendix	32
2.9.1 Site selection for soil thermal regime monitoring	
2.9.2 Thaw depth monitoring.	
2.9.3 Air and soil temperature monitoring	
2.9.4 Vegetation Surveys	
2.9.5 Image acquisition and preparation for supervised peatiand classification.	
2.9.0 Supervised peatiand classification	24
2.9.7 Trajectories of young merinokarst bog development	
2.9.8 Scaling of regional peat plateau loss due to thermokalst expansion	55
Chanter 3. Effects of wildfire on soil respiration in normafrost postlands of the Ca	nadian
Western Boreal Forest	56
3.1 Abstract	

3.1 Abstract	
3.2 Introduction	

3.3 Methods	
3.3.1 Study area	59
3.3.2 Field data collection	
3.3.3 Modeling and statistical analysis	61
3.4 Results.	62
3.4.1 Soil temperature.	62
3.4.2 Soil respiration.	63
3.4.3 Soil respiration models	63
3.4.4 Deep peat contribution to CO <sub>2</sub> release	64
3.5 Discussion.	64
3.6 Conclusion	67
3.7 Acknowledgements	68
3.8 References.	69
Chapter 4: Conclusions	84
4.1 Literature Cited	86
Bibliography	87

# List of Tables

**Table 3.2**: Coefficients (±1SE) of variables in the best model based on an AIC evaluation.....80

# List of Figure

**Fig S2.3.** Determination of talik presence, and estimation of thaw depth at point locations where taliks were absent. A) Example of seasonal development of the thawed soil layer at 6 representative point locations at the Lutose 2012 site. Soil probe was 150 cm long, hence measures of 150 cm indicate that no frost table was reached. We classified point locations as having taliks if there was a >90 cm increase in thaw depth between two monitoring occasions, as exemplified by three point locations. Only a subset of sites had thaw depths monitored as frequently as Lutose 2012, and thus for other sites we did not have direct observations of talik frequency. B) Histogram of September point location depth to frost table measurements (n = 100) from the Fort Simpson 1995 site overlaid by a gaussian kernel density estimation (Wessa 2015). All point locations with September thaw depths >100 cm were assumed to have taliks, an assumption that was tested against observed talik presence at a few sites (See table S2 and S3). We used the maximum density to estimate representative active layer depth of point locations where taliks were absent, as using either the mean or median thaw depths would be strongly influenced by talik frequency.

#### **Chapter 1: Introduction**

#### 1.1 Overview of the Importance of Permafrost Peatlands

Peatlands only cover 3-5% of the world's land area but store  $\sim$ 30% of terrestrial soil carbon [*Gorham 1991*]. Within northwestern Canada, permafrost is found almost exclusively in peatlands in the southern limit of the discontinuous permafrost zone. Approximately 28% of peatlands in continental western Canada are underlain by permafrost and store  $\sim$ 13 Pg of soil carbon. Given that peatlands 1) currently serve as a large terrestrial carbon sink and 2) are located primarily at northern latitudes that are undergoing rapid and accelerated warming associated with climate change, understanding their response to disturbances is vital for continued benefits derived from these ecosystems.

Peatlands within the discontinuous permafrost zone not only serve as a large terrestrial carbon sink, they also supply cultural resources such as traditional medicines, travel corridors, serve as headwaters for numerous river systems and help maintain stable wildlife populations [*Callahan et al., 2005, Walsh et al., 2005*]. When these systems undergo thawing events that can lead to either rapid thaw settlement or thermokarst development it can result in the ground instability which has important implications for land processes and hazards such as surface hydrology, groundwater regimes, and stability. These changes have further reaching consequences for socio-economics of northern communities, infrastructure and development projects [*Savo et al., 2017*].

It is predicted that by the mid-21<sup>st</sup> century, the area of permafrost in the northern hemisphere will decline by 20-35 per cent [*IPCC 2016*]. Permafrost peatlands near the southern extent of the permafrost zone are most likely to be impacted first, thereby triggering a cascade of potentially detrimental effects for the global carbon cycle and ecosystem services derived from permafrost peatlands. Therefore it is important to focus research efforts on understanding mechanisms that may trigger or accelerate permafrost thaw in order to make informed management and development decisions in the future.

#### 1.2 Permafrost in Western Canada and Development of Peatland Complex

Across western Canada 21%, or over 365,000km<sup>2</sup>, of the land is covered by peatlands [*Vitt et al.*, 2000]. In this region, peatlands are a heterogeneous mosaic of peatland types including ombrotrophic bogs, minerotrophic fens, and permafrost features (peat plateaus). Permafrost underlays approximately 28% of these peatlands [*Tarnocai et al.*, 2004] and is located within the discontinuous permafrost zone where it is restricted mostly to ombrotrophic peatlands. In these areas peat plateaus arise because the rate of production of organic matter exceeds losses from decomposition, disturbance, and leaching [*Vitt et al.*, 1995] and as the result of being ice rich, these features are elevated 0.5-1.5m above surrounding wetlands [*Quinton et al.*, 2011]. These areas are distinguishable by the presence of black spruce (*Picea mariana*) that is able to establish due to the creation of an aerobic active layer in the upper part of the ground due to summer seasonal thaw.

Many studies have shown that on century time scales both the aggradation and degradation of plateaus can occur within the same peatland [e.g. *Zoltai*, 1972; *Payette et al.*, 1976]. Permafrost is maintained in peatlands due to a hydrological gradient that exists between forested plateaus and surrounding wetlands (bogs and fen) such that adjacent wetlands receive relatively large inputs of runoff from plateaus [*Quinton et al.*, 2009]. The ground surface on peat plateaus is dominated by bryophytes and the insulating properties provided by dry surface peat, mosses and lichens are vital for the persistence of permafrost in peatlands. In the winter, given thin enough snow layers, heat is lost through the frozen peat and in the summer, permafrost is insulated from increasing air temperatures due to the presence of *Sphgnaum fuscum* hummocks and *Cladonia rangiferina* [*Zoltai*, 1993], which reduces the heat flux downward towards the permafrost [*Seppala 1988*].

However, degradation of plateaus can be triggered by both natural and catastrophic events. As ice accumulates, a plateau will continue to increase in height to a point where the peat can no longer cover the surface [*Seppala 1988*]. Once this occurs, surface cracks begin to form, creating deep fissures allowing warm air and rain to reach the frozen core and initiate thawing [*Zoltai*, 1972]. As thawing progresses, the land subsides, increasing the saturation of the surface peat as it begins to fall into or below the rooting zone. Black spruce, the dominant tree species on peat plateaus, cannot tolerate these waterlogged, low-oxygen conditions [*Islam and Macdonald*, 2004] in the rooting zone and begin to die. Loss of black spruce is accompanied by *Sphagnum fuscum*, characteristic of the dry peat plateaus, being gradually replaced by *Sphagnum* 

*angustifolium* then by *Sphagnum riparium* as subsistence continues and the water table approaches the surface of the peat [*Gignac et al.*, 1991; *Zoltai*, 1993]. This thawing process, under some circumstances, can be initiated by wildfire but historically it was believed that fire did not cause widespread permafrost thaw in peatlands [*Thie*, 1974] and that the rate of aggregation and degradation of permafrost were in balance [*Zoltai*, 1993]. This process is referred to as thermokarst formation and while it is natural, it is predicted that our changing climate may accelerate rates of thermokarst formation. Over the latter half of the 20<sup>th</sup> century, the area of permafrost peatlands decreased by 22% on average [*Beilman and Robinson*, 2003]. This conversion of permafrost plateaus to permafrost-free bog and fens will have drastic implications for vegetation succession, carbon storage, and ecosystem services derived from these areas [*Turetsky et al.*, 2007].

Permafrost thaw can also occur in the regions in the form of top-down thaw. The active layer, the layer of soil between the permafrost and the surface that thaws each summer due to increased air temperatures, can become progressively deeper each summer. In some circumstances, seasonal thaw becomes so great that due to the quick onset of winter not all of the energy stored in the thawed soil is able to be evacuated. This can result in a perennially unfrozen peat layer between the permafrost table and the frozen upper layers called a talik. The soils within a talik remain above 0°C thereby acting as a year-round heat source for further active layer deepening [*Viereck et al.*, 2008].

## **1.3 Wildfire and the Boreal Forest**

Although not historically common, peat stratigraphy and macrofossil analysis has shown evidence of some wildfire-induced permafrost thaw; a transition from silvic peat (decomposed woody material) to *Sphagnum* mosses is preceded by a char layer. Wildfire is a major stand-renewing agent for much of the circumboreal forest zone and is an integral part in determining patterns of vegetation and forest structure within the boreal [*Rowe and Scotter*, 1973; *Stocks et al.*, 2002]. Wildfire within peatlands represents a large source of carbon to the atmosphere, burning over 1500km<sup>2</sup> and releasing 6GtC annually in western Canada alone [*Turetsky et al.*, 2002] on a 100-120 year fire return interval [*Turetsky et al.*, 2004]. Peatlands have been considered relatively resistant to wildfire due to their moist soil conditions [*Kuhry*, 1994] but recent studies have shown that soil combustion in peatlands during wildfire can be substantial

both in extent [*Turetsky et al.*, 2004] and severity [*Turetsky and Wieder*, 2001; *Turetsky*, 2002; *Benscoter and Wieder*, 2003; *Turetsky et al.*, 2011]. Wildfire affects forested peat plateaus in similar frequencies to uplands forests (100-120 fire return interval; *Turetsky et al. 2004*) and in the last 50 years more than 25% of peat plateaus within the Taiga Plains have burned. Changes to the fire regime are expected to occur in the future under a predicted changing climate that could drastically alter fire dynamics on the landscape and potentially increase these numbers. Climate change has altered the boreal fire regime and these changes are predicted to become more pronounced in the future [*Kasischke and Turetsky*, 2006; *Flannigan et al.*, 2009]. Across North America, the total annual burn area has increased due to higher mean annual temperatures and drier conditions [*Kasischke and Turetsky*, 2006]. Fire occurrence is currently predicted to increase by 25% by the year 2030 and 75-150% by the end of the century [*Wotton et al.*, 2010].

#### 1.4 Interaction between Wildfire and Permafrost

Wildfire is an important driver of ecosystem structure. Within the discontinuous permafrost zone, by combusting aboveground biomass and some of the uppermost organic material, wildfire can dramatically alter the land surface energy balance by exposing soils and reducing the insulation effect of organic layers. This causes modifications to the ground thermal and hydrological conditions that can affect permafrost stability [Yoshikawa et al., 2002]. Several studies have begun to explore the effects of wildfire on permafrost stability and have noted impacts from 1 to more than 40 years after fire [e.g. Burn, 1998; Yoshikawa et al., 2002; Viereck et al., 2008; Brown et al., 2015; Zhang et al., 2015]. Studies show that fire can increase soil temperatures for several decades and the active layer usually deepens [Mackay, 1995; Burn, 1998; Viereck et al., 2008; Rocha et al., 2012; Nossov et al., 2013]. Some observations show permafrost table stabilization or recovery 5-25 years after a fire [Mackay, 1995; Yoshikawa et al., 2002; Smith et al., 2015]; others show steady deepening of the permafrost table three to six decades after a fire or removal of vegetation and surface organic matter [Burn, 1998; Viereck et al., 2008] and even complete degradation of permafrost after a fire [Viereck et al., 2008]. Several modeling studies have shown that vegetation recovery and the accumulations of organic material are important for stabilization and recovery of the permafrost table [e.g. Jafarov and Romanovsky, 2013; Smith et al., 2015].

There is sufficient evidence to suggest wildfire plays a role in permafrost conditions,

however, presently the effects of fire are not considered in traditional permafrost maps [eg. Brown et al. 1997, Heginbottom et al. 1995]. Furthermore, global- and national-scale permafrost modelling studies do not consider the effects of fire, often due to their coarse spatial resolution [Zhang, 2005; Koven et al., 2015]. Although fine scale studies have recognized the importance of fire on soil thermal conditions and show significant increases in near-surface soil temperatures and active layer thickening, they have not made the connection between potential active layer thickening and thermokarst development. These two phenomena have only been explored in isolation. Furthermore, most of the research conducted has been done in Alaska where peat depth is often <1m thick and on north-facing hillsides rather than in larger lowland peatland complexes [e.g. Viereck et al., 2008; Brown et al., 2015], much less than that found in the western boreal peatlands of Canada. These studies suggest that the response of permafrost to wildfire is mediated by fire severity (% organic layer consumption). For example, Nossov et al. (2013) argue that when less than 4cm of organic material remains post fire it is likely to lead to complete permafrost loss. However, the vulnerability of permafrost to wildfire is likely to differ across the discontinuous permafrost zone depending on numerous site factors that influence soil heat flux. The current literature neglects the wetland rich areas of northwestern Canada where permafrost is found under slightly raised (0.5-1.0m), tree covered plateaus that overlay frozen organic soils up to and >10m in depth [Wright et al., 2009]. Understanding permafrost response to wildfire in this region is important as it contains over 20% of the circumpolar soil organic carbon stock and approximately a quarter of these permafrost soils have burned in the last 30 years.

### 1.5 Carbon Cycling in the Boreal

Northern peatland ecosystems accumulate carbon because annual net primary productivity (NPP) of peatland vegetation generally exceeds lateral transport and the annual decomposition of litter and peat. Over millennia NPP has been greater than decomposition and hence, northern peatlands have been a persistent sink of CO<sub>2</sub>, averaging 0.02 to 0.03 kg CO<sub>2</sub>-C  $m^{-2} y^{-1}$  [*Gorham*, 1991; *Tolonen and Turunen*, 1996] . One of the most significant knowledge gaps related to the northern permafrost regions is the impact thawing of frozen soils will have on the global carbon cycle. Some researchers believe the effect may be catastrophic, while some are skeptical about its significance. Understanding the mechanisms that control aerobic and

anaerobic respiration pre-thaw and post-thaw at different spatial and temporal scales is particularly important for predicting the effects of changing disturbance regimes and a changing climate on northern carbon cycling.

Following permafrost thaw, herein referring to active layer deepening for purposes of this thesis, large quantities of labile carbon, which were previously resistant to decomposition due to subzero temperatures, become available for decomposition, resulting in the transfer of old carbon from the terrestrial environment to the atmosphere. The rate and timing of this, however, remains a key uncertainty [Schuur et al., 2015]. Variations in post thaw trajectories (active layer deepening versus thermokarst creation) can drive differences in CO<sub>2</sub> production rates due to differences in oxygen conditions [Schuur and Bockheim, 2008]. In contrast to thermokarst formation that results in anaerobic conditions, active layer deepening typically exposes thawed soil organic carbon stores to aerobic conditions [Lee et al., 2012]. These aerobic conditions are suspected to be particularly important in controlling the rate of CO<sub>2</sub> transfer to the atmosphere because decomposition that uses oxygen as the terminal electron acceptor is 5 to 10 time faster than anaerobic decomposition [Bridgham et al., 1998]. Furthermore, rates of soil respiration can further be expected to increase as soil temperatures post fire are greater due to changes in the surface energy budget that results from the removal of the vegetation layer and exposure of a predominantly black charred surface [Randerson et al., 2006]. This increase in soil temperature is particularly important as in peat, carbon dioxide production rates increase by a factor of 2-3 for every 10°C temperature increase (Q<sub>10</sub>) [Moore and Dalva, 1993; Yavitt et al., 1997; *McKenzie et al.*, 1998].

Decomposition of thawed permafrost soils, whether due to thermokarst formation or active layer deepening, will result in the decomposition of old carbon that had been previously protected from decomposition. Previous work examining the release of old carbon following permafrost thaw have shown that following thermokarst development up to 15% of ecosystem respiration is derived from the decomposition of old carbon [*Schuur and Bockheim*, 2008]. However, most work examining the decomposition of previously frozen carbon stores has examined rates of decomposition following thermokarst development. Much less studied is the potential for old carbon losses associated with deepening active layers, particularly following wildfire. Up to a 200% increase in active layer depth has been observed following fire [*O'Neill et al.*, 2002], representing a vast pool of organic matter that is now available for microbial

decomposition and could represent a large pulse of carbon to the atmosphere from the terrestrial environment. The magnitude and timing of this decomposition is not well understood. Furthermore, previous studies that have attempted to understand this have focused primarily on black spruce upland forests, which tend to have shallower organic layers, and examined net ecosystem respiration. While this provides a good approximation for trends in  $CO_2$  production following a wildfire, it provides little insight into the source of carbon. It is unclear if observed increases in ecosystem respiration are due to increased soil temperatures in the upper peat surfaces or due to loss of old carbon at depth.

#### **1.6 Research Objectives**

The overall goal of my research is to document the temporal and spatial time frame over which wildfire affects permafrost stability and carbon cycling in the discontinuous permafrost zone. Despite previous studies showing changes to the surface energy balance post fire [*Randerson et al.*, 2006; *Thompson et al.*, 2015] and deepening of active layers [*Viereck et al.*, 2008; *Brown et al.*, 2015], no studies have explored the link between these factors and widespread permafrost thaw. Using a chronosequence approach (space for time substitution), I aim to understand the spatial extent of wildfire induced permafrost thaw and the fate of soil carbon post-fire by asking the following questions:

- 1. Over what temporal scale does wildfire affect active layer depths?
- 2. To what extent does fire accelerate thermokarst development in discontinuous peatlandrich regions?
- 3. How is soil respiration affected over the years to decades following wildfire in peat plateaus?
- 4. Relative to unburned peat plateaus, what is the magnitude of post-fire CO<sub>2</sub> respiration and what proportion of this respiration results from the decomposition of deeper, recently thawed C stores?

#### 1.5.1 Wildfire and Permafrost Stability

Preliminary investigations have explored the impacts of wildfire on active layer deepening in Alaska where organic layer depths are often <1m. In chapter 2, I determine if and to what extent fire induced active layer deepening leads to thermokarst formation. Active layer

depths were measured across a series of sites that burned from 1967-2014. These sites were also fitted with several temperature loggers to monitor seasonal variation in soil temperatures with time since fire. This allowed me to assess the temporal scale over which wildfire impacts active layer depths in peat rich areas such as the Taiga Plains. I was then able to use high resolution satellite imagery with supervised classification to determine if the area of thermokarst varies between adjacent burn and unburn areas. As a result of wildfire on the surface energy balance, and therefore the soil thermal regime, I expect that 1) active layer depths will be deeper during the first decade, followed by a decade of widespread talik formation, 2) soil temperatures at depth will be greater due to a greater ground heat flux, and 3) as a consequence of (1) and (2) thermokarst formation will be accelerated within a burned area compared to an adjacent unburned area.

#### 1.5.2 Post-Fire Carbon Cycling

Changes in carbon emissions post thaw has largely been associated with increased methane respiration due to the creation of anaerobic conditions within thermokarst features and the respiration of old carbon [*Schuur and Bockheim*, 2008; *Schuur et al.*, 2015]. Chapter three deals with changes in carbon cycling caused by the deepening of the aerobic zone (active layer) due to wildfire. Using  $CO_2$  fluxes from dark static chambers on an unburn site and sites that burned in 2000 and 2007 allowed me to document the magnitude of soil respiration both temporally and in comparison to an unburned site. Because wildfire increases the availability of soil organic carbon and increases the temperatures of the soils, I expect that 1) soil respiration in burned sites will be greater compared to unburned sites, and 2) burned sites will have proportionally more respiration that is derived from deeper soils due to higher soil temperatures at depth.

Chapter four will summarize the most important findings, analyze the strengths and weaknesses of this study, and provide direction for future research in relationship to wildfire and permafrost stability. Integrating the information contained with chapters two and three, I provide a discussion on possible future permafrost-fire-carbon scenarios in a changing climate.

#### **1.7 Literature Cited**

Beilman, D., and S. Robinson (2003), Peatland permafrost thaw and landform type along a climatic gradient, *Proc. 8th Int. Conf. Permafr.*, 61–65.

- Benscoter, B. W., and R. K. Wieder (2003), Variability in organic matter lost by combustion in a boreal bog during the 2001 Chisholm fire, *Can. J. For. Res.*, 33(12), 2509–2513, doi:10.1139/x03-162.
- Bridgham, S. D., K. Updegraff, and J. Pastor (1998), Carbon, nitrogen, and phosphorus mineralization in northern wetlands, *Ecology*, 79(5), 1545–1561, doi:10.1890/0012-9658(1998)079[1545:CNAPMI]2.0.CO;2.
- Brown, D. R. N., M. T. Jorgenson, T. A. Douglas, V. E. Romanovsky, K. Kielland, C. Hiemstra, E. S. Euskirchen, and R. W. Ruess (2015), Interactive effects of wildfire and climate on permafrost degradation in Alaskan lowland forests, *J. Geophys. Res.*, 120, 1619–1637, doi:10.1002/2015JG003033.Received.
- Brown, J., O.J. Ferrians, Jr., J.A. Heginbottom, and E.S. Melnikov, eds. 1997. Circum-Arctic map of permafrost and ground-ice conditions. Washington, DC: U.S. Geological Survey in Cooperation with the Circum-Pacific Council for Energy and Mineral Resources. Circum-Pacific Map Series CP-45, scale 1:10,000,000, 1 sheet.
- Burn, C. R. (1998), The active layer: Two contrasting definitions, *Permafr. Periglac. Process.*, 9(4), 411–416.
- Callaghan, T.V., Bjorn, L.O., Chapin III, S.F.; Chernov, Y., Christensen, T.R., Huntley, B., Ims, R., Johansson, M., Riedlinger, D.J., Jonasson, S. Matveyeva, N., Oechel, W., Panikov, N., and Shaver, G. 2005. Arctic Tundra and Polar Desert Ecosystems. In Arctic Climate Impact Assessment, Cambridge University Press, Cambridge, United Kingdom.
- Flannigan, M., B. Stocks, M. Turetsky, and M. Wotton (2009), Impacts of climate change on fire activity and fire management in the circumboreal forest, *Glob. Chang. Biol.*, 15(3), 549– 560, doi:10.1111/j.1365-2486.2008.01660.x.
- Gignac, L. D., D. H. Vitt, S. C. Zoltai, and S. E. Bayley (1991), Bryophyte Response Surfaces

Along Climatic, Chemical, and Physical Gradients in Peatlands of Western Canada, *Nov. Hedwigia*, *53*(1–2), 27–71.

- Gorham, E. (1991), Northern Peatlands: Role in the Carbon Cycle and Probable Responses To Climatic Warming, *Ecol. Appl.*, *1*(2), 182–195.
- Heginbottom, J.A., and M.A. Dubreuil. 1995. Canada -- Permafrost in *National Atlas of Canada*, 5th edition, Plate 2.1 (MCR 4177), scale 1:7,500,000.
- Islam, M. A., and S. E. Macdonald (2004), Ecophysiological adaptations of black spruce (Picea mariana) and tamarack (Larix laricina) seedlings to flooding, *Trees Struct. Funct.*, 18(1), 35–42, doi:10.1007/s00468-003-0276-9.
- IPCC, 20016: Climate Change 20016: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M.Tignor and H.L. Miller (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Jafarov, E., and V. Romanovsky (2013), The effects of fire on the thermal stability of permafrost in lowland and upland black spruce forests of interior Alaska in a changing climate, *Environ. Res. Lett.*, 8(3), doi:10.1088/1748-9326/8/3/035030.
- Kasischke, E. S., and M. R. Turetsky (2006), Recent changes in the fire regime across the North American boreal region—Spatial and temporal patterns of burning across Canada and Alaska, *Geophys. Res. Lett.*, 33(9), L09703, doi:10.1029/2006GL025677.
- Koven, C. D., D. M. Lawrence, and W. J. Riley (2015), Permafrost carbon climate feedback is sensitive to deep soil carbon decomposability but not deep soil nitrogen dynamics, *Proc. Natl. Acad. Sci.*, *112*(12), 3752–3757, doi:10.1073/pnas.1415123112.
- Kuhry, P. (1994), of Sphagnum- in the development The role of fire boreal Canada in western peatlands dominated, *J. Ecol.*, 82(4), 899–910.
- Lee, H., E. A. G. Schuur, K. S. Inglett, M. La Voie, and J. P. Chanton (2012), The rate of permafrost carbon release under aerobic and anaerobic conditions and its potential effects on

climate, Glob. Chang. Biol., 18, 515-527, doi:10.1111/j.1365-2486.2011.02519.x.

- Mackay, J. R. (1995), Active Layer Changes (1968 to 1993) Following the Forest-Tundra Fire near Inuvik, N. W. T., Canada, Arct. Alp. Res., 27(4), 323–336.
- McKenzie, C., S. Schiff, R. Aravena, C. Kelly, and V. St. Louis (1998), Effect of temperature on production of CH4 and CO2 from peat in a natural and flooded boreal forest wetland, *Clim. Change*, 40, 247–266, doi:10.1023/A:1005416903368.
- Moore, T. R., and M. Dalva (1993), The influence of temperature and water table position on carbon dioxide and methane emissions from laboratory columns of peatland soils, *J. Soil Sci.*, 44(4), 651–664, doi:10.1111/j.1365-2389.1993.tb02330.x.
- Nossov, D. R., M. T. Jorgenson, K. Kielland, K. D. Johnson, J. W. Harden, A. D. Mcguire, G. Grosse, S. Goetz, and A. D. Mcguire (2013), The effects of fire on the thermal stability of permafrost in lowland and upland black spruce forests of interior Alaska in a changing climate, *Environ. Res. Lett.*, (8), 1–11, doi:10.1088/1748-9326/8/3/035030.
- O'Neill, K. P., E. S. Kasischke, and D. D. Richter (2002), flux following fire in black spruce, white spruce, and aspen stands of interior Alaska, *Can. J. For. Res.*, *32*(9), 1525–1541, doi:10.1139/x02-077.
- Payette, S., H. Samson, and D. Lagarec (1976), The evolution of permafrost in the taiga and in the forest-tundra, wester Quebec-Labrador Peninsula, *Can. J. For. Res.*, 6(2), 203–220, doi:10.1007/s13398-014-0173-7.2.
- Quinton, W. L., M. Hayashi, and L. E. Chasmer (2009), Peatland Hydrology of Discontinuous Permafrost in the Northwest Territories: Overview and Synthesis, *Can. Water Resour. J.*, 34(4), 311–328, doi:10.4296/cwrj3404311.
- Quinton, W. L., M. Hayashi, and L. E. Chasmer (2011), Permafrost-thaw-induced land-cover change in the Canadian subarctic: implications for water resources, *Hydrol. Process.*, 25(1), 152–158, doi:10.1002/hyp.7894.
- Randerson, J. T. et al. (2006), The Impact of Boreal Forest Fire on Climate Warming, *Science* (80-. )., 314(5802), 1130–1132, doi:10.1126/science.1132075.

- Rocha, A. V, M. M. Loranty, P. E. Higuera, M. C. Mack, F. S. Hu, B. M. Jones, A. L. Breen, E.
  B. Rastetter, S. J. Goetz, and G. R. Shaver (2012), The footprint of Alaskan tundra fires during the past half-century: implications for surface properties and radiative forcing, *Environ. Res. Lett.*, 7(4), 44039, doi:10.1088/1748-9326/7/4/044039.
- Rowe, J., and G. W. Scotter (1973), Fire in the boreal forest, *Quat. Res.*, *3*, 444–464, doi:10.1016/0033-5894(73)90008-2.
- Seppälä, M. 1988. Palsas and related forms. In: Clark, M.J. (ed.). Advances in Periglacial Geomorphology, pp. 247–278. John Wiley and Sons, Chichester
- Savo, V., D. Lepofsky, H. Bay, and H. Bay (2017), *Impacts of Climate Change on Subsistence-Oriented Communities*, Elsevier Inc.
- Schuur, E., and J. Bockheim (2008), Vulnerability of permafrost carbon to climate change: Implications for the global carbon cycle, *Bioscience*, 58(September), 701–714, doi:10.1641/B580807.
- Schuur, E. A. G. et al. (2015), Climate change and the permafrost carbon feedback, *Nature*, *250*, 217–227, doi:10.1038/nature14338.
- Smith, S. L., D. W. Riseborough, and P. P. Bonnaventure (2015), Eighteen Year Record of Forest Fire Effects on Ground Thermal Regimes and Permafrost in the Central Mackenzie Valley, NWT, Canada, *Permafr. Periglac. Process.*, 26, 289–303, doi:10.1002/ppp.1849.
- Stocks, B. J. et al. (2002), Large forest fires in Canada, 1959–1997, J. Geophys. Res., 108, doi:10.1029/2001JD000484.
- Tarnocai, C., F. Mark Nixon, and L. Kutny (2004), Circumpolar-Active-Layer-Monitoring(CALM) sites in the Mackenzie Valley, northwestern Canada, *Permafr. Periglac. Process.*, 15(2), 141–153, doi:10.1002/ppp.490.
- Thie, J. (1974), Distribution and thawing of permafrost in the southern part of the discontinuous permafrost zone in Manitoba, *Arctic*, *27*(3), 189–200.

Thompson, D. K., A. S. Baisley, and J. M. Waddington (2015), Seasonal variation in albedo and

radiation exchange between a burned and unburned forested peatland: implications for peatland evaporation, *Hydrol. Process.*, 29(14), 3227–3235, doi:10.1002/hyp.10436.

- Tolonen, K., and J. Turunen (1996), Accumulation rates of carbon in mires in Finland and implications for climate change, *The Holocene*, 6(2), 171–178, doi:10.1177/095968369600600204.
- Turetsky, M. (2002), Current disturbance and the diminishing peatland carbon sink, *Geophys. Res. Lett.*, *29*(11), 7–10, doi:10.1029/2001GL014000.
- Turetsky, M. R., and R. K. Wieder (2001), A direct approach to quantifying organic matter lost as a result of peatland wildfire, *Can. J. For. Res.*, *31*(2), 363–366, doi:10.1139/x00-170.
- Turetsky, M. R., R. K. Wieder, and D. H. Vitt (2002), Boreal peatland C fluxes under varying permafrost regimes, *Soil Biol. Biochem.*, 34(7), 907–912, doi:10.1016/S0038-0717(02)00022-6.
- Turetsky, M. R., B. D. Amiro, E. Bosch, and J. S. Bhatti (2004), Historical burn area in western Canadian peatlands and its relationship to fire weather indices, *Global Biogeochem. Cycles*, 18(4), 1–9, doi:10.1029/2004GB002222.
- Turetsky, M. R., R. K. Wieder, D. H. Vitt, R. J. Evans, and K. D. Scott (2007), The disappearance of relict permafrost in boreal North America: Effects on peatland carbon storage and fluxes, *Glob. Chang. Biol.*, *13*, 1922–1934, doi:10.1111/j.1365-2486.2007.01381.x.
- Turetsky, M. R., E. S. Kane, J. W. Harden, R. D. Ottmar, K. L. Manies, E. Hoy, and E. S. Kasischke (2011), Recent acceleration of biomass burning and carbon losses in Alaskan forests and peatlands, *Nat. Geosci.*, 4(1), 27–31, doi:10.1038/ngeo1027.
- Viereck, L. A., N. R. Werdin-pfisterer, and P. C. Adams (2008), Effect of Wildfire and Fireline Construction on the Annual Depth of Thaw in a Black Spruce Permafrost Forest in Interior Alaska : A 36-Year Record of Recovery, *Ninth Int. Conf. Permafr.*, 1845–1850.
- Vitt, D. H., Y. Li, and R. J. Belland (1995), Patterns of bryophyte diversity in peatlands of continental western Canada, *Bryologist*, *98*(2), 218–227, doi:10.2307/3243306.

- Vitt, D. H., L. a Halsey, I. E. Bauer, and C. Campbell (2000), Spatial and temporal trends in carbon storage of peatlands of continental western Canada through the Holocene, *Can. J. Earth Sci.*, 37(5), 683–693, doi:10.1139/e99-097.
- Walsh, J.E., Anisimov, O.H., Hagen, J.O.M., Jakobsson T., Oerlemans, J., Prowse, T.D., Romanovsky, V., Savelieva, N., Serreze, M., Shiklomanov, A., Shiklomanov, I. and Solomon, S. 2005. Cryosphere and Hydrology, In Arctic Climate Impact Assessment, Cambridge University Press, Cambridge, United Kingdom.
- Wotton, B. M., C. a. Nock, and M. D. Flannigan (2010), Forest fire occurrence and climate change in Canada, *Int. J. Wildl. Fire*, *19*, 253–271, doi:10.1071/WF09002.
- Wright, N., M. Hayashi, and W. L. Quinton (2009), Spatial and temporal variations in active layer thawing and their implication on runoff generation in peat-covered permafrost terrain, *Water Resour. Res.*, 45(5), 1–13, doi:10.1029/2008WR006880.
- Yavitt, J. B., C. J. Williams, and R. K. Wieder (1997), Production of methane and carbon dioxide in peatland ecosystems across North America: Effects of temperature, aeration, and organic chemistry of peat, *Geomicrobiol. J.*, 14(September), 299–316, doi:10.1080/01490459709378054.
- Yoshikawa, K., W. R. Bolton, V. E. Romanovsky, M. Fukuda, and L. D. Hinzman (2002), Impacts of wildfire on the permafrost in the boreal forests of Interior Alaska, J. Geophys. Res., 108(D1), 16–17, doi:10.1029/2001JD000438.
- Zhang, T. (2005), Influence of the Seasonal Snow Cover on the Ground Thermal Regime : an Overview, *Rev. Geophys.*, 43, 1–23, doi:10.1029/2004RG000157.1.INTRODUCTION.
- Zhang, Y., S. A. Wolfe, P. D. Morse, I. Olthof, and R. H. Fraser (2015), Journal of Geophysical Research : Earth Surface, J. Geophys. Res. Earth Surf., (120), 2338–2356, doi:10.1002/2015JF003679.
- Zoltai, S. C. (1972), Palsas and Peat Plateaus in Central Manitoba and Saskatchewan, *Can. J. For. Res.*, 2(3), 291–302, doi:10.1139/x72-046.
- Zoltai, S. C. (1993), Cyclic Development of Permafrost in the Peatlands of Northwestern

Canada, Arct. Alp. Res., 25(3), 240-246, doi:1df0.2307/1551820.

#### Chapter 2: Wildfire as a major driver of recent permafrost thaw in boreal peatlands

#### 2.1 Abstract

Permafrost vulnerability to climate change may be underestimated unless effects of wildfire are considered. In this study we assessed decadal impacts of wildfire on soil thermal regime and abrupt permafrost thaw in western Canadian boreal peatlands. Using 16 sites that vary in time since fire, we found a near doubling in the depth to frost table in the year following fire while continuously thawed soil layers above the permafrost, taliks, subsequently became ubiquitous a decade after fire. The soil thermal regime slowly recovered with the reestablishment of black spruce and lichens, and effects of wildfire could not be distinguished at sites that burned >30 years ago. Given the perturbation of the soil thermal regime, we hypothesized accelerated abrupt permafrost thaw along permafrost edges, i.e. thermokarst bog expansion, during the first three decades following wildfire. To test this we quantified extents of developmentally young thermokarst bogs in four large peatlands partially affected by fire in the 1980s or 1990s. Young thermokarst bogs were nearly twice as abundant within fire scars, indicating a tripling in the rate of expansion after fire. Our results suggest over the last 30 years, wildfire is directly responsible for 2,200  $\pm$  1,500 km<sup>2</sup> (95%CI) of thermokarst bog expansion, and thus responsible for ~25% of all recent permafrost thaw in this region. Permafrost thaw has major implications for traditional land-use, regional hydrology and water quality, and soil greenhouse gas emissions, and our study stresses the recent and future role of wildfire for permafrost stability in circumpolar boreal regions.

#### 2.2 Significance statement

A quarter of the global land surface has permanently frozen ground, permafrost. Ongoing climate warming is causing rapid permafrost thaw, which has important effects on infrastructure integrity, traditional land-use, soil carbon storage and more. In this study we show that permafrost thaw in boreal regions is significantly accelerated in the first three decades following a wildfire. We estimate that wildfire is directly responsible for ~25% of all permafrost thaw over the last 30 years in a region in western Canada. Hence, projections of future permafrost thaw are likely underestimates unless they incorporate the effects of wildfire.

#### **2.3 Introduction**

Permafrost thaw has impacts at local to global scales, affecting infrastructure integrity (1), traditional land-use (2, 3), wildlife habitat (4), hydrology, and water quality (5, 6), and soil greenhouse gas emissions (7). Projections of 21<sup>st</sup> century permafrost stability indicate particularly rapid thaw in the discontinuous permafrost zone (8, 9), but projections may yet be underestimates as they do not explicitly consider potential destabilization of permafrost due to the role of disturbances, the largest of which in the boreal forest is wildfire. Wildfire affects 2.5 Mha of the boreal biome on average each year (10). Peat plateaus are ubiquitous in the boreal forest, and often burn preferentially due to the black spruce trees. Peat stratigraphy in boreal peatlands show that historical wildfire's effects on soil thermal regimes or the spatial extents of permafrost thaw arising due to wildfire. Understanding the role of wildfire as a driver of permafrost thaw is critical given the recent, and projected, increases in annual burned areas in boreal regions (12).

This study focuses on the discontinuous permafrost zone within the Taiga Plains ecozone in western Canada, a 400,000 km<sup>2</sup> region that is representative of circumpolar boreal regions with extensive peatland cover (Fig. 2.1a). Here, peat accumulation initiated following deglaciation ~9000 years ago (13), and current peat depths are between 2 and 6 m. As such, this region has some of the highest soil organic carbon concentrations globally (14). Permafrost started aggrading after the Holocene thermal maximum ~5,000 years ago, and became more widespread following a climate cooling ~1,200 years ago (15). Permafrost is largely restricted to peatlands in the study region, due to the insulating properties of peat and the relatively warm mean annual temperatures between -1°C and -6°C. A relatively warm climate, and shallow permafrost basal depths less than 20 m (16), suggest that even minor perturbations to the surface energy balance can cause complete and rapid thaw.

Peatlands in the study region are a fine-scale mosaic dominated by permafrost-affected peat plateaus and non-permafrost thermokarst bogs (Fig 2.1b). Peat plateaus are elevated 1 to 3 m above their surrounding peatlands due to excess ground ice, causing generally dry surface conditions and vegetation dominated by black spruce (*Picea mariana*), Labrador tea shrubs (*Rhododendron groenlandicum*), and lichens (*Cladina spp.*) (Fig. 2.2). Black spruce forests burn preferentially in boreal western Canada (17), and ~ 25% of peat plateaus in the study region have

burned in the last 30 years (Fig. 2.1a and Fig S2.1). Thermokarst bogs form through abrupt permafrost thaw, often developing from small depressions on peat plateaus which then expand radially and coalesce. Abrupt permafrost thaw cause peat surface collapse, which initially leads to a young thermokarst bog stage with saturated soils and vegetation dominated by *Sphagnum riparium* and sedges (*Carex aquatilis*). Vegetation composition shifts to *Sphagnum fuscum* and ericaceous shrubs as new peat accumulates and the surface becomes drier (Fig 2.1b and Fig 2.2, *SI Appendix*, Fig. S2.1). Repeat aerial photography and satellite image acquisition have shown that the rate of thermokarst bog expansion has accelerated in the last few decades due to warming (18–20), but no analysis has been done to compare rates of thermokarst bog expansion within and outside historical fire scars.

The objective of this study was to assess impacts of wildfire on permafrost stability in peat plateaus. We hypothesize that wildfire causes decadal perturbations to the surface energy balance, resulting in higher soil temperatures and a thicker active layer. We further hypothesize that the presence of a seasonally unfrozen layer during the winter further perturbs the soil thermal regime which renders peat plateau edges increasingly vulnerable to abrupt permafrost thaw through thermokarst bog expansion (Fig. 2.2). In order to test our hypotheses, we monitored soil thermal regimes at 16 peat plateau sites, ten of which had burned at various times in the last 50 years (Fig. 2.1a and Fig 2.3, *Supplementary Appendix*, Table S2.1), and we assessed extents of recently developed young thermokarst bogs in four large peatlands partially affected by fire 20 to 30 years ago (*Supplementary Appendix*, Table S2.2). By combining field measurements and remote sensing approaches, we are able to estimate the area and relative importance of permafrost thaw in the study region directly occurring as a consequence of recent wildfires.

#### 2.4 Results and Discussion

# 2.4.1 Effects of wildfire on peat plateau soil thermal regimes

Soil thermal regimes at each of the 16 peat plateau sites (Fig 2.1a, *Supplementary Appendix*, Table S2.1) were characterized by monitoring air and soil temperatures at 10 and 40 cm depths for a full year and by repeat measurements of depth to frost table up to 7 times per site between May and late September of thawed soil layer depths at 100 point locations in a 5 m spaced grid (see methods). By tracking thaw depth development throughout the season, it was

apparent from sudden large increases in thaw depths that many point locations had continuously thawed soil layers between the permafrost and the seasonally frozen layer, i.e. taliks (Fig. 2.2, *Supplementary Appendix*, Fig S2.3a). Using the within-site distribution of depth to frost table measurements in late September, we derived two indices for characterizing site soil thermal regimes; the proportion of point locations underlain by taliks, and the maximum depth to frost table for point locations not underlain by taliks (*Supplementary Appendix*, Fig S2.3b). All sites except one had peat depths > 1.9m, uniform pre-fire tree density, and mean annual air temperatures among the six unburned sites were between 0.8 to  $-3.1^{\circ}$ C (*Supplementary Appendix*, Fig S2.4 and Table S2.1), hence differences in soil thermal regimes are considered likely to be primarily due to differences in fire histories.

We observed nearly 50% deeper depths to the frost table in recently burned sites, with  $\sim$ 50 cm in unburned sites compared to  $\sim$ 85 cm in sites burned in the last 5 years (Fig. 2.4a). We associate this phenomenon predominantly to a faster thaw rate throughout the summer (Supplementary Appendix, Fig S2.5a and Fig S2.5b), as opposed to difference in timing of spring thaw initiation or late season thawing dynamics, though these factors likely play a small role as well. This was supported by seasonal thaw at 40 cm depth occurring two weeks earlier in recently burned sites compared to unburned sites (Fig 2.4b). More rapid and deeper development of the seasonally thawed layer likely stems primarily from a perturbed surface energy balance with increased incident shortwave radiation following the loss of the black spruce canopy and reduced albedo due to the replacement of white lichen with black char (Fig 2.3 and Fig 2.4c). Interestingly, in southern boreal permafrost free bogs, wildfire causes reduced soil moisture that insulates the soil profile thereby counteracting the effects of albedo and increased incident shortwave radiation (25). However, permafrost peat plateaus already exhibit low surface moisture such that the effect of wildfire on near surface moisture and thermal conductivity is likely much more limited. These observed effects of wildfire on the seasonal cumulative ground heat flux, as indicated by end of season depths to the frost table, were strongest in the first 5-10 years following fire, and then faded and disappeared 20 to 30 years after the fire.

We also observed that wildfire has led to substantial expansion of taliks and increased soil temperatures at depth, yet these effects were delayed and were most pronounced 10 to 20 years following the fire (Fig 2.4a and Fig 2.4b). During this time period 72-100% of a burned site was underlain with a talik, compared to ~20% of unburned sites (Fig 2.4a). A similar pattern

was found for maximum seasonal soil temperatures at 40cm depth for which burned sites exceeded that of the unburned sites by as much as 4°C, and ultimately reached a maximum temperature of 10.7° C, 16 years post-fire (Fig 2.4b). This is in contrast to southern permafrost free bogs where post fire, deep soils remain buffered from changes in surface soil temperatures (25). The development of taliks during this time period results from there not being enough heat loss during the winter to re-freeze the seasonally thawed layer. This is likely caused by both a greater heat loss requirement for burned sites and reduced heat loss during the winter in burned sites as a result of deeper snow due to the loss of an intercepting canopy. We interpret these delayed effects on talik coverage and maximum soil temperatures at 40cm as the cumulative multi-year effects of a perturbed surface energy balance. Following the disappearance of the short term effects (ie. increased depths to the frost table and increased thaw rates, 20-30 years post fire) is when these longer term effects start to recover (Fig 2.4a and Fig 2.5b) where by the taliks start to refreeze and soil temperatures decrease.

We observed no apparent influence of fire severity on the trajectories of the soil thermal regime following wildfire. Fire severity, as measured by depth of burn into the organic layer, has been shown to potentially affect successional trajectories in other boreal ecosystems (26). The ten burned sites varied with regards to fire size and seasonality (Supplementary Appendix, Table S2.1), which both can affect severity, yet neither measure influenced the observed soil thermal regime nor vegetation recovery. Given that peat plateau fires are predominantly crown fires, and in our sites fire only burned off a small fraction of the peat, we suspect there is likely little influence on the recruitment of black spruce seedlings or understory vegetation. This is important as both of these factors are known to be strongly influenced by deep organic layer removal during high severity fires (27, 28). Furthermore, we find that even a low severity fire will cause tree mortality (loss of intercepting canopy) and the loss of lichen, as indicated by the loss of tree canopy and complete lack of lichen species in sites that burned <20 years ago (Fig 2.3.). Given that we attribute loss of lichen and an intercepting canopy as the major driver of the perturbed energy balance (which results regardless of the severity of the fire) we do not expect a large influence of fire severity on the energy imbalance and subsequently increasing depths to the frost table.

The recovery of the soil thermal regime is likely linked to vegetation recovery, which strongly regulates the surface energy balance (29). Recovery of the seasonally thaved layer

following fire coincides with the recovery of the lichen (*Cladonia spp.*) and black spruce trees (Fig 2.4c) Recovery of talik coverage and deep soil temperatures take an additional 10 years before they cannot be noticed at all. This is indicative that these factors represent the inter-annual cumulative effects on the energy balance. Wildfire likely affects the surface energy balance during both winter, spring, and summer. In the winter the loss of an intercepting canopy can cause a thickening of snowpack which increases insulation (30), while in the spring decreased albedo due to the loss of lichen can increase convective heat transport and thus affects snow ablation (31), and in the summer the lack of shading and decreased albedo (29) causes increased energy transfer through the soils. Taking these alterations to the energy balance into account, we created a conceptual model that identifies an important link between the vertical energy balance and lateral thaw rates that result in thermokarst bog expansion. We propose that for permafrost to persist in peat plateaus, which are surrounded by warmer unfrozen peat soils, it is necessary that there is a substantial heat loss to the atmosphere in the winter, since there is net heat gain both from below the permafrost and from surrounding non-permafrost bogs. We suspect that if an unfrozen soil layer persists, there can be no heat loss from the permafrost core to the atmosphere during the winter and this will result in a net increase in energy of the permafrost core until the soil thermal regime has recovered (i.e. 30 years). Consequently, this causes increased thaw in the form of young thermokarst bog development (Fig 2.2).

#### 2.4.2 Effects of wildfire on young thermokarst bog expansion

Recently developed young thermokarst bogs along thawing peat plateau edges have distinct vegetation composition that is clearly discernable as bright green in high resolution satellite imagery (Fig. 2.1 and Fig. 2.5). Radiocarbon dating of thermokarst bog peat cores from the study region have shown that dominance of *Sphagnum riparium* and *Carex aquatilis* lasts for between 60 and 140 years following permafrost thaw before peat accumulation causes a succession to dominance of rust-colored *Sphagnum fuscum* and ericaceous shrubs (15) (*Supplementary Appendix*, Table S2.2). We took advantage of these ecological shifts to determine the long term effects of wildfire on permafrost thaw by assessing the spatial extents of peat plateaus, young thermokarst bogs and mature thermokarst bogs within four large peatlands (Fig 2.1). The four chosen peatlands were partially affected by wildfires 20-30 years ago (*Supplementary Appendix*, Table S2.3), i.e. the period over which field data showed that they

were perturbed and thus higher rates of thermokarst bog expansion can be expected. Within each peatland, we interpreted the areal extents of young thermokarst bogs as indicators of permafrost thaw over the last 100 years, i.e. a period which includes and extends beyond the potential effects of fire.

Supervised classification using high resolution satellite imagery was carried out for 250 × 250 m peatland sections that were visually deemed to only contain peat plateaus, young and old thermokarst bogs (*Supplementary Appendix*, Fig S2.6) (see methods). At least 34, 250 × 250m cells for each burned and unburned peatlands site were classified, where the burned and the unburned peatland locations were located <20km apart. Using validation transects of the young thermokarst - mature thermokarst bog transition edge and the young thermokarst bog – peat plateau edges (*Supplementary Appendix*, Fig S2.7) we found young thermokarst bog area to generally be underestimated in the burned site and overestimated in the unburned site (*Supplementary Appendix*, Fig S2.8). Overall precision was found to be that 80% of the time the classification edges were within 1 m of the field delineated edges. This suggests that measured difference in young thermokarst bog area between the burned and unburned area is not due to difference in the classification of these two areas and we therefore deem the classification of young thermokarst to be both precise and reliable indicator of recent permafrost thaw difference between burned and unburned areas (last 100 years).

The area of young thermokarst bogs were on average ~65% greater within 20- to 30-yearold fire scars compared to nearby unburned peatland sections. Average area coverage of young thermokarst bogs were 8.8% in burned and 5.9% in the unburned areas (Fig 2.6a.). Variation in the amount of young thermokarst bog abundance among the unburned sites was likely driven by variable mean annual temperatures. The lowest young thermokarst bog abundance (~2.5% peatland coverage) was found for the Zama and Trout Lake sites, located at elevations ~300 m higher than the other sites, while the lower elevation Fort Simpson site (5% peatland coverage) is located ~250 km north of the 60<sup>th</sup> parallel site (10%). While an absolute effect of fire at +3% was similar across sites, the relative importance of fire for permafrost thaw was greater at the colder sites, compared with with 30% more young thermokarst bog in warm sites compared 250% more young thermokarst bog in colder sites.

The effect of wildfire on young thermokarst bog development rates is however, greater than indicated by current coverage of thermokarst bogs, since much of the young thermokarst bogs currently present in peatlands likely developed before fires that occurred 20-30 years ago (Fig 2.6b). Estimating differences in rates of young thermokarst development between burned and unburned peatland sections in the decades after wildfire requires a set of assumptions (*Supplementary Appendix*, Fig S2.9) of thermokarst development. In brief, we assumed that young thermokarst bogs currently present in peatlands started developing sometime between 50 and 150 years ago (15) (*Supplementary Appendix*, Table S2.2), and that it is likely that rates of thermokarst bog expansion has increased in the last 30 years in unburned peatlands due to a warming climate (18–20). Based on our assumptions, we estimate that the rate of young thermokarst bog development has nearly tripled within burned peatlands compared to nearby unburned peatlands, at  $0.28 \pm 0.11$  and  $0.10 \pm 0.07$  % yr<sup>-1</sup> (95% CI), respectively (Fig 2.6b).

Development rates of young thermokarst bogs can also be expressed as rates of peat plateau loss. Across the four classified peatland sites, coverage of peat plateaus varied between 50 and 75% (*Supplementary Appendix*, Table S2.3). As such, our estimated rates of peat plateau loss in the last 30 years were  $0.39 \pm 0.18$  % yr<sup>-1</sup> and  $0.16 \pm 0.12$  % yr<sup>-1</sup> (95% CI), respectively for burned and unburned peatlands. Our rate of peat plateau loss due to thermokarst bog expansion within unburned areas is thus similar but in the low range of what other studies have estimated used historical image change detection (0.26-0.34 % yr<sup>-1</sup> peat plateau loss) (18, 19). By combining the distribution of peat plateaus, the distribution and timing of fires (Fig 2.1), and our estimated rates of peat plateau loss within and outside burned peatlands, we estimate that the total permafrost loss over the last 30 years within the study region is 7,600 ± 4,100 km<sup>2</sup> (95% CI) (See supplementary information). We estimate that 2,200 ±1,500 km<sup>2</sup> of peat plateau thaw is directly attributed to wildfire, i.e. ~23% of the total permafrost loss. As such, wildfire has been a major driver of recent permafrost thaw in the western boreal forest of Canada.

#### **2.5** Conclusion

In this study, we assessed the impacts of wildfire on peat plateau soil thermal regimes and young thermokarst bog expansion to provide a spatial estimate of permafrost thaw attributed to recent fires across the Taiga Plains discontinuous permafrost zone in northwestern Canada. Our findings, in conjunction with earlier research (32–34), highlight the importance of wildfire as a driver of permafrost thaw. This is particularly pressing given the known effect of climate change on wildfire activity within the boreal region (35–37). With larger fire years burning later into the

season it is predicted that greater areas of permafrost soils will burn and, based on our findings this will trigger widespread thermokarst formation, as we have show that wildfire leads to a near tripling in the rate of recent thermokarst formation for ~30 years post fire. As the circumpolar regions continue to warm at twice that of the global average (38) these ecosystems will be increasingly stressed and it is unlikely that permafrost will regenerate at the rate at which we are losing it. Areas that have undergone thermokarst formation represent a potentially important player in the positive feedback loop as recently formed thermokarst features are known to have the potential to cause large methane emission (39, 40). If we apply our estimated 25% thaw due to wildfire on methane emission from recent thermokarst features it suggests that potentially 0.055 Gg CH<sub>4</sub> yr<sup>-1</sup> has been released to the atmosphere simply from land cover shifts associated with recent thermokarst formation just in this study region alone.

Globally these peatland permafrost features are highly significant, covering 8% of the circumpolar world and residing within the boreal forest that is subjected to wildfire every 100-120 years. Our study region is representative of boreal wetland thermokarst landscapes (41), therefore our results will help us understand global impacts of wildfire on permafrost stability.

#### 2.6 Materials and Methods

#### 2.6.1 Site selection for soil thermal regime monitoring

Field data for this study was collected at representative sites within the discontinuous permafrost zone of the Taiga Plains (Fig. 2.1a). Within the study area, permafrost peat plateaus comprise 33% of the landscape, storing over 24 Pg of carbon in the upper 300cm. In the past 30 years, 25% of these permafrost peat plateaus have been affected by wildfire. Six locations were chosen in areas with one unburned site and at least one historically burned site that vary in time since fire (2, 3, 4, 9, 10, 15, 21, 33, 40, 48 years ago) (Fig. 2.1, *Supplemental Appendix*, Table S2.1). Each of these sites contained at least 2 m of organic peat soil (*Supplemental Appendix*, Table S2.1). Unburned sites did not differ in mean annual air temperature (*Supplemental Appendix*, Table S2.1) and had similar vegetation composition, which consisted principally of stunted black spruce and small evergreen shrubs. Lichen and feather mosses dominated areas of less defined microtopography whereas *Sphagnum spp*. generally dominated hummocks and hollows, which covered <20% of a given site. More details in *SI Appendix*.

#### 2.6.2 Depth to frost table monitoring

In each site 100 permanently marked points were established in a 10 x 10 grid with 3 m spacing centrally located on the peat plateau (*Supplemental Appendix*, Fig S2.1). At each point we performed repeated measures of thaw depth up to 150 cm and recorded the microtopography (hummock, hollow, undefined) and dominant ground cover (lichen, *Sphagnum spp.*, feathermoss, char). Hollows were excluded from the analysis as they are known to experience hydrological convergence which results in deeper thaw depths due to higher heat conductance (16) (*Supplemental Appendix*, Table S2.1). Depth to frost table was measured in early June, mid-July, and early-September. At the 2012, 2008, 2000 and their paired unburned site depth to frost table was measured 7 times through the summer to allow for determination of locations with taliks. More details in *SI appendix*.

#### 2.6.3 Air and soil temperature monitoring

In two locations in each site, an Onset Hobo 8k Pendant logger was installed in a lichen/char covered area at 10cm and 40cm depth. An additional logger was placed under a white shade on a pole ~1m in height at each site to monitor air temperatures. Loggers recorded temperatures every 2 hours for a year from September 2015 – September 2016, and were then aggregated for a daily average. More details in *SI appendix*.

#### 2.6.4 Vegetation Surveys

During August 2016, vegetation analysis was completed using the point intercept method (42) along three 30 m transects in each site. Tree heights of all black spruce trees within a meter on each side of the transects was visually estimated. More details in *SI appendix*.

#### 2.6.5 Image acquisition and preparation for supervised peatland classification

Multispectral WorldView2 images were obtained for four sites within the study area (Fig 2.1a). All images were taken between June and September in 2011. At each location, two  $5\times5$  km images were acquired, one including burned areas and one adjacent unburned area. Within the acquired  $5\times5$  km satellite image there were substantial areas that were unsuitable for analysis as they contained fens, uplands, and lakes. To examine the influence of wildfire on permafrost
stability solely in peatlands, 500 sections ( $250 \times 250$  m) were formed from the original 5x5 km image (*Supplementary Appendix*, Fig S2.6). Cells that contained only peat plateaus, young and mature thermokarst bogs were retained for further analysis. This resulted in 57-61 cells, depending on the location of image (*Supplementary Appendix*, Table S2.3). More details in *SI appendix*.

## 2.6.6 Supervised peatland classification

Using our field knowledge we defined three spectrally separable land covers types: peat plateaus, young thermokarst bog, and mature thermokarst (Fig 2.5). Maximum likelihood classification was performed in ENVI based on the spectral signatures of these three classes. More details in *SI appendix*.

## 2.6.7 Accuracy assessment of peatland classification

Classification validation transects were collected in August 2016 at the Zama burned and Zama unburned location using a dGPS unit accurate to 10cm. The user defined edge of the young thermokarst bog to mature thermokarst bog edge and the young thermokarst bog to peat plateau edge were walked and GPS points collected approximately every 2 meters (*Supplementary Appendix*, Fig S2.7). To assess classification accuracy the distance between the user collected field edge and the classification edge was determined (*Supplementary Appendix*, Fig S2.8.).

## 2.6.8 Estimating trajectories of young thermokarst development

A range of possible trajectories based on timing of recent thermokarst bog development, percent thaw per year as a result of fire, at each site were estimated using a boot strapping approach ( $\sim$ 5000 iterations). Such estimations require a set of assumptions detailed in *SI appendix*.

## 2.6.9 Spatial scaling of peat plateau loss due to thermokarst development

To scale peat plateau thaw to the scale of the study region we applied our estimated rate of young thermokarst development per year to the total burn area within the study area. More details in *SI Appendix*.

## 2.7 Acknowledgements

We thank Liam Heffernan, Katheryn Burd, McKenzie Kuhn, Michael Barbeau and Jessi Steinke for assistance with field work. We greatly acknowledge financial support for the research from the following sources: National Science and Engineering Research Council Discovery grant (RGPIN-2016-04688), the Campus Alberta Innovates Program, the University of Alberta Northern Research Awards, and the Northern Scientific Training Program.

## **2.8 References**

- 1. Melvin AM, et al. (2016) Climate change damages to Alaska public infrastructure and the economics of proactive adaptation. *Proc Natl Acad Sci* 114(2):E122–E131.
- 2. Andrews TD, et al. (2016) Permafrost Thaw and Aboriginal Cultural Landscapes in the Gwich'in Region, Canada. *APT Bull J Preserv Technol* 47(1):15–22.
- 3. Hovelsrud GK, Poppel B, Van Oort B, Reist JD (2011) Arctic societies, cultures, and peoples in a changing cryosphere. *Ambio* 40:100–110.
- 4. Woo M, Young K, Brown L (2006) High Arctic Patchy Wetlands: Hydrologic Variability and Their Sustainability. *Phys Geogr* 27(4):297–307.
- 5. Frey KE, McClelland J w. (2009) Impacts of permafrost degradation on arctic river biogeochemistry. *Hydrol Process* 23:169–182.
- Olefeldt D, Roulet NT (2014) Permafrost conditions in peatlands regulate magnitude, timing, and chemical composition of catchment dissolved organic carbon export. *Glob Chang Biol* 20(10):3122–3136.
- 7. Schuur EAG, et al. (2015) Climate change and the permafrost carbon feedback. *Nature* 250:217–227.
- 8. Lawrence DM, Slater AG, Swenson SC (2012) Simulation of present-day and future permafrost and seasonally frozen ground conditions in CCSM4. *J Clim* 25(7):2207–2225.
- 9. Zhang Y, Chen W, Riseborough DW (2008) Transient projections of permafrost distribution in Canada during the 21st century under scenarios of climate change. *Glob*

*Planet Change* 60:443–456.

- Stocks BJ, et al. (2002) Large forest fires in Canada, 1959–1997. J Geophys Res 108(August 2002):1–12.
- Zoltai SC (1993) Cyclic Development of Permafrost in the Peatlands of Northwestern Canada. Arct Alp Res 25(3):240–246.
- Kasischke ES, Turetsky MR (2006) Recent changes in the fire regime across the North American boreal region—Spatial and temporal patterns of burning across Canada and Alaska. *Geophys Res Lett* 33(9):L09703.
- Loisel J, et al. (2014) A database and synthesis of northern peatland soil properties and Holocene carbon and nitrogen accumulation. *The Holocene* 24(9):1028–1042.
- 14. Hugelius G, et al. (2014) Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. *Biogeosciences* 11(23):6573–6593.
- Pelletier N, et al. (2017) Influence of Holocene permafrost aggradation and thaw on the paleoecology and carbon storage of a peatland complex in northwestern Canada. *The Holocene*:1–15.
- Quinton WL, Hayashi M, Chasmer LE (2011) Permafrost-thaw-induced land-cover change in the Canadian subarctic: implications for water resources. *Hydrol Process* 25(1):152–158.
- Cumming SG (2001) Forest Type and Wildfire in the Alberta Boreal Mixedwood : What Do Fires Burn. *Ecol Appl* 11(1):97–110.
- Baltzer JL, Veness T, Chasmer LE, Sniderhan AE, Quinton WL (2014) Forests on thawing permafrost: Fragmentation, edge effects, and net forest loss. *Glob Chang Biol* 20(3):824–834.
- 19. Chasmer LE, Hopkinson C (2016) Threshold loss of discontinuous permafrost and landscape evolution. *Global Biogeochem Cycles* doi: 10.11:1–15.
- 20. Mamet SD, Chun KP, Kershaw GGL, Loranty MM, Kerhaw GP (2017) Recent Increases

in Permafrost Thaw Rates and Areal Loss of Palsas in the Western Northwest Territories, Canada. *Permafr Periglac Process*. doi:10.1002/ppp.1951.

- 21. Brown J, Ferrians O., Heginbottom J., Melnikov ES (1998) *Circumarctic map of permafrost ground ice conditions* (Boulder, CO).
- 22. Ecosystem Classification Group (1995) A National Ecological Framework for Canada.
- Hugelius G, et al. (2013) The Northern Circumpolar Soil Carbon Database: spatially distributed datasets of soil coverage and soil carbon storage in the northern permafrost regions. *Earth Syst Sci Data* 5:3–13.
- 24. Natural Resource Canada (2017) Canadian Wildland Fire Information System. Available at: http://cwfis.cfs.nrcan.gc.ca/datamart/metadata/nfdbpoly [Accessed July 14, 2017].
- Kettridge N, Thompson DK, Waddington JM (2012) Impact of wildfire on the thermal behavior of northern peatlands: Observations and model simulations. J Geophys Res Biogeosciences 117. doi:10.1029/2011JG001910.
- Johnstone JF, Hollingsworth TN, Chapin FS, Mack MC (2010) Changes in fire regime break the legacy lock on successional trajectories in Alaskan boreal forest. *Glob Chang Biol* 16(4):1281–1295.
- 27. Johnstone JF, Chapin FS (2006) Effects of soil burn severity on post-fire tree recruitment in boreal forest. *Ecosystems* 9(1):14–31.
- 28. Gibson CM, et al. (2016) Variation in plant community composition and vegetation carbon pools a decade following a severe fire season in interior Alaska. *J Veg Sci*:1–11.
- Randerson JT, et al. (2006) The impact of boreal forest fire on climate warming. *Science* 314(5802):1130–1132.
- Hedstrom NR, Pomeroy JW (1998) Measurements and modeling of snow interception in the boreal forest. *Hydrol Process* 12:1611–1625.
- 31. Gleason KE, Nolin AW, Roth TR (2013) Charred forests increase snowmelt: Effects of burned woody debris and incoming solar radiation on snow ablation. *Geophys Res Lett*

40:4654-4661.

- Brown DRN, et al. (2015) Interactive effects of wildfire and climate on permafrost degradation in Alaskan lowland forests. J Geophys Res G Biogeosciences 120(8):1619–1637.
- 33. Viereck LA, Werdin-pfisterer NR, Adams PC (2008) Effect of Wildfire and Fireline Construction on the Annual Depth of Thaw in a Black Spruce Permafrost Forest in Interior Alaska : A 36-Year Record of Recovery. *Ninth Int Conf Permafr*:1845–1850.
- Zhang Y, Wolfe SA, Morse PD, Olthof I, Fraser RH (2015) Journal of Geophysical Research : Earth Surface. *J Geophys Res Earth Surf* (120):2338–2356.
- Flannigan M, et al. (2013) Global wildland fire season severity in the 21st century. For Ecol Manage 294:54–61.
- Amiro BD, Cantin a., Flannigan MD, de Groot WJ (2009) Future emissions from Canadian boreal forest fires. *Can J For Res* 39(2):383–395.
- Kasischke ES, Turetsky MR (2006) Recent changes in the fire regime across the North American boreal region - Spatial and temporal patterns of burning across Canada and Alaska. *Geophys Res Lett* 33(9):1–5.
- Chapin FS, Sommerkorn M, Robards MD, Hillmer-Pegram K (2015) Ecosystem stewardship: A resilience framework for arctic conservation. *Glob Environ Chang* 34:207–217.
- Olefeldt D, Turetsky MR, Crill PM, Mcguire a. D (2013) Environmental and physical controls on northern terrestrial methane emissions across permafrost zones. *Glob Chang Biol* 19(2):589–603.
- Walter KM, Zimov SA, Chanton JP, Verbyla D, Chapin FS (2006) Methane bubbling from Siberian thaw lakes as a positive feedback to climate warming. *Nature* 443(7107):71–75.
- 41. Olefeldt D, et al. (2016) Circumpolar distribution and carbon storage of thermokarst

landscapes. Nat Commun 7:13043.

42. Jonasson G (1983) The point intercept method for non-destructive estimation of biomass. *Phytocoenologia* 11(3):385–388.

## 2.9 SI Appendix

## 2.9.1 Site selection for soil thermal regime monitoring

The Northern Circumpolar Soil Carbon Database (version 2; (1)) was used to determine the area of permafrost peat plateaus within the study area. The database provides spatially distributed quantification of soil organic carbon at 0-300 cm depths and soil type across the circumpolar permafrost region. Burned area from the past 30 years from the Canadian National Fire Data (2) base was overlaid. Proportion histel burned was determined by multiplying area of burned polygon by proportion of histels within given area.

## 2.9.2 Thaw depth monitoring

A Kruskal-Wallace one way analysis of variance for non-parametric data was performed to identify difference between points within a grid at a given location based on landcover and topography. At all site hollows displayed significantly deeper depths to frost table (p < 0.05), regardless of landcover and were therefore excluded from analysis.

Depth to frost table was determined using a kernel density estimator - a nonparametric technique way to estimate the probability density function of a random variable (3) as it is more representative of the ecological dynamics than mean or median (*Supplementary Appendix*, Fig S4). By using the max density estimator (4) we were able to identify locations where there are no underground hydrological convergence points (5). A kernel density estimation allowed us to more accurately represent the depth to frost table in locations where taliks are absent, as using either a mean or median thaw depth would strongly influence our estimation of active layer depth based on talik frequency.

Talik distribution was determined using end of thawing season (September) measurements of depth to frost table. Locations with thaw depths greater than 100cm were deemed to be underlain with a talik (Fig S2.3). To verify this we performed an error matrix with sites where thaw depths had been more frequently monitored. From May to July, thaw depths in the 2012 burn sites and its paired unburn site were measured four times over the course of the early summer (May-July). We manually identified locations of taliks as places where thaw depths doubled between two consecutive sampling periods. We then set an approximation that end of season thaw depths that were 90cm  $\pm$  40% would have a talik. We tested this against our

manually identified locations of taliks and performed an error matrix. Our accuracy assessment for predictions of absence or presence of taliks within a burned peat plateau site using September thaw depths showed 92% accuracy in the unburned site and 97% in the burned site (Table S2.4 and Table S2.5).

## 2.9.3 Air and soil temperature monitoring

Soil temperature loggers were installed in areas of undefined topography in lichen or char to avoid areas of increased conduction due to hollows or the presence of *Sphagnum spp*. Soil temperatures were averaged between the two loggers in each site. During time periods where there was logger failure, logger data from the remaining functioning logger only was used.

#### 2.9.4 Vegetation Surveys

Vegetation surveys were completed by delineating a 50 m baseline transect along a randomly chosen geographic bearing. Three 30 m transects were established perpendicular to the 50m transect: one at the center at the 25 m mark and two on either side of the center at randomly generated distances greater than 5 m away from the center point. Understory species abundance was measured using the point intercept method (6) along all three transects with a pin being dropped every meter. Species abundance was calculated using the following formula: Species abundance (% cover) = (number of hits per species/total number of hits)\*100. Tree heights were visually estimated within 1 m of either side of each 30 m transect for a total sampling area of  $180m^2$ .

## 2.9.5 Image acquisition and preparation for supervised peatland classification

The commercial WorldView2 satellite provides 46 cm panchromatic resolution with three visible bands within the visible spectrum (blue 450 - 520 nm; green 520 - 600 nm; red 625 - 695 nm).

#### 2.9.6 Supervised peatland classification

Within the WorldView2 imagery plateaus are characterized by relatively high albedo *Cladonia* spp., speckled with relatively low albedo black spruce. Bogs are characterized by spectrally bright *Sphagnum* spp. and thermokarst by spectrally bright *Sphagnum riparium* and graminoids (Fig 2.5). Post classification sieve and clumping was done to remove small numbers of pixels within larger contiguous classed and to achieve a smoother overall classification.

#### 2.9.7 Trajectories of young thermokarst bog development

A range of possible trajectories of young thermokarst bog development at each site were estimated with the following assumptions of the boot strap exercise. 1) Sections that burned 30 years ago had identical trajectory as unburned sections up until the time of fire. 2) The young thermokarst bogs that we see today on the landscape started developing  $100 \pm 50$  years ago (95%) CI), as indicated by <sup>14</sup>C and Pb dating of peat cores (Table S2.2). 3) We assume that the upper limit of the 95% CI range for young thermokarst bog development rate before the fire event would be similar as if there has been no increase in rate in the last 30 years due to a warming climate. However, ample evidence exists to suggest that thermokarst bog expansion have increased as a result of climate warming. We therefore deem it highly unlikely that that thermokarst expansion rates in unburned sections were higher prior to the time of the fire as compared to after. We define this rate as current median thermokarst extent in unburned sections divided by years since currently present thermokarst bogs started developing (i.e. age from distribution described in assumption 2). 4) We assume that the lower limit of the 95% CI range for young thermokarst bog development rate before the fire is half of the rate calculated from the upper limit 95% CI development rate before the fire, i.e. current median thermokarst extent in unburned sections divided by years since currently present thermokarst bogs started developing divided by 2.

Simply, it is unlikely that there had been no thermokarst development in the period before the fire. It is also unlikely that the rates of young thermokarst bog development were higher in the period before the fire compared to after for unburned sections. Our central estimate of the amount of young thermokarst bog on the landscape at the time of the fire is half of present day coverage, which is consistent with several studies that suggest a doubling in the rate of thermokarst development in the past 30 years (7–9). Following the fire event, we assume a

constant rate of young bog development that led to currently present observed young thermokarst bog coverage (Fig S2.9).

This analysis was completed for each location and the average and standard deviation of mean development rates across the four sites was used to create our estimated trajectory of young thermokarst development (Fig 2.6b).

## 2.9.8 Scaling of regional peat plateau loss due to thermokarst expansion

Using the Northern Circumpolar Soil Carbon Database (version 2; (1)) and the Canadian National Fire Data (1) we determined the area of histel burned each year for the past 60 years. We assume fire affected areas thaw at an increased rate for 30 years. Therefore, observed present day young thermokarst bogs are the result of fires that occurred prior to 1980s whose effects are beginning fade and transition to mature thermokarst, and post 1980s whose effects are only beginning to be seen. We applied the increased fire-induced rate of thaw to the histel area burned within a given year times the number of years since fire, or pre-30 years ago. From this derived total thaw area due to fire. Total thaw area due to climate warming alone was derived by multiplying total histel area within study area by our estimate of thaw from the unburned sites. Our central estimate of thaw rate and standard deviation was determined using a bootstrapping approach (~300 iterations) to determine fire-induced thaw area and climate induced thaw area.



**Fig. 2.1.** Wildfire in western Canadian boreal peatlands. A) Outline of the study region, defined as the discontinuous permafrost zone (21) within the Taiga Plains ecozone (22). Shading within the study region shows the coverage of peat plateaus as indicated by histel soils (23), while historical burned areas from the last 30 years are shown as hashed areas (24). Yellow stars indicate locations where field data characterizing soil thermal regimes were collected, with one unburned and at least one burned site monitored at each star location. Blue stars indicate locations where satellite image classification of recent thermokarst bog expansion was carried out within extensive peatlands partially affected by wildfire. B) Example of a peatland characteristic for the study region, with three open non-permafrost thermokarst bog stage is visible as green areas along some peat plateau edges, a result of recent permafrost thaw. Photo copyright Mason Stothart. C) Example of a characteristic crown fire burning a peat plateau, near Fort Simpson, NWT, in 2013. Photo copy right Franco Alo.



**Fig. 2.2.** Simplified illustration of soil thermal states for peat profiles along transects from thermokarst bogs to peat plateaus within and outside historical burned areas.



Fig. 2.3. Vegetation recovery at select peat plateau sites used in this study to assess long term effects of fire on soil thermal regime.



**Fig. 2.4.** Trajectories of soil thermal regime and vegetation recovery on peat plateaus following wildfire. A) Trajectories of depth to frost table at point without taliks, and proportion of points where thaw depth thaw monitored that had continuously thawed soil layers, i.e. taliks. No depths to frost table estimates were possible for sites burned 7 and 16 years ago due to 100% talik coverage. B) Trajectories rate of that, as indicated by soils at 40cm rising above 0.5°C, and maximum seasonal soil temperature attained at 40 cm depth (points at 7 and 49 years post fire not available). C) Trajectories of vegetation recovery, exemplified by the percent ground cover of lichens, and median height of black spruce trees. In each panel, the average of 6 unburned sites is shown on the left with grey background, where error bars indicate  $\pm 1$  standard deviation for percent lichen and interquartile range for black spruce trees.



**Fig 2.5.** Classification of peat plateau, young thermokarst bog, and mature thermokarst bog extents in large peatlands using high resolution satellite imagery (World View 2, 50 cm resolution). A) and C) Supervised classification was carried out for 250 x 250 m sections of unburned and burned peatlands respectively that were visually assessed to only contain peat plateau, young thermokarst bog, and mature thermokarst bogs. Sections containing channel fens, upland forests, ponds, clouds, or shadows were excluded from analysis and are shown with grey shading. B) and D ) Example of a 250 × 250 m section selected for classification in the unburned and burned peatlands respectively.



**Figure 2.6.** Effect of wildfire on permafrost thaw through development of young thermokarst bogs in boreal western Canada. A) Current-day coverage of young thermokarst bogs within and outside 20- to 30-year-old fire scars at four large peatlands. Box plots indicate median, interquartile range, and minimum and maximum young thermokarst bog coverage among classified  $250 \times 250$  m sections, and number of classified sections is indicated (n). In each pair, burned sections had significantly greater was significantly different from its unburned pair (p < 0.01, pairwise t-test). Sites are ordered left to right by decreasing mean annual air temperature. B) Temporal development ( $\pm 95\%$  CI) of young thermokarst bogs persist in peatlands for  $100 \pm 40$  years before succession into mature thermokarst bogs (15), suggesting that a significant proportion of the currently present young thermokarst bogs formed prior to the historical wildfire. Rate of development of young thermokarst bog, expressed as a % of total peatland area ( $\pm 95\%$  CI), shown above each trajectory.

# **SI Figures and Tables**

**Table S2.1**: Summary statistics from chronosequence sites. Date and size of fire derived from Large Fire Data Base (NRCAN 2016). Number of active layer measurements include only locations used in analysis, ie. hollows excluded.

Site	Site Coordinates	Elevation (masl)	Date Fire	of	Size of Fire (ha)	Peat Depth (cm)	Thaw depth point locations (n)	Mean Annual Air Temp <sup>1</sup> (°C)
Samba Deh 1967	61°21'18"N 121° 0'29"W	237	June 1967	20,	605	>250	100	-2.89
Camsell 1975	62°17'59"N 122°38'8"W	293	June 1975	24,	32,087	230	92	-2.34
Zama 1984	59°20'45"N 119°20'24"W	587	July, 1982	10	6132	260	92	N/A <sup>2</sup>
Fort Simpson 1995	61°59'45"N 121°23'45"W	186	Augus 18,199	t 95	4653	>300	100	-0.15
Lutose 2000	59°30'0"N 117°12'17"W	318	June 2000	28,	1471	280	100	-0.58
Fort Simpson 2006	61°59'24"N 121°23'51"W	185	July 2006	22,	808	200	100	-0.14
Lutose 2007	59°25'2"N 117°17'4"W	324	June 2007	13,	2245	>300	100	-0.91
Lutose 2012	59°46'42"N 117° 3'40"W	299	July 2012	9,	37,244	>300	100	N/A <sup>2</sup>
Samba Deh 2013	61°11'42"N 120° 5'25"W	290	June 2013	27,	96,330	290	99	-1.82
Kakisa 2014	60°56'26"N 117°21'42"W	225	May 2014	26,	80,311	255	100	-1.91
Camsel Unburn	62°14'46"N 122°34'23"W	262	-		-	130	99	-2.51
Samba Deh Unburn	61°11'59"N 120° 6'50"W	287	-		-	290	98	-2.04
Lutose Unburn	59°29'4"N 117°10'41"W	305	-		-	>300	95	-1.45
Zama Unburn	59°22'9"N 119°19'16"W	584	-		-	260	100	-3.1
Kakisa Unburn	61° 4'53"N 117°37'23"W	250	-		-	200	82	-1.57
Fort Simpson Unburn	61°59'34"N 121°22'59"W	186	-		-	>300	96	-0.81

 $^{1}$ July 2015 – 2016

<sup>2</sup>Temperature sensor malfunctioned

**Table S2.2.** Dating of transitions between plateau peat and young thermokarst bog peat in cores collected from current young and mature thermokarst bog sites. All dating is done  $\sim$ 1-2 cm below the transition, which means that true transition dates happened more recently than indicated by the dating techniques. Note that these cores were not collected at the transition between young and mature thermokarst bogs, but rather centrally within young and mature thermokarst bogs, respectively, and thus in order to assess the persistence of young thermokarst bogs in the landscape, we are interested in seeing the maximum age of transitions within current young thermokarst bogs and the minimum age of transition in current mature thermokarst bogs.

Core type and ID	Location	Depth of transition (cm)	Age of transition (cal YBP)	Transition dating method	Reference
Young thermokar	st bog cores				
SC1-E	61°18'N 121°18'W	0	25	Aerial photo	Pelletier et al. 2017
SC1-YB	61°18'N 121°18'W	61	130	<sup>210</sup> Pb	Pelletier et al. 2017
SC2-YB	61°18'N 121°18'W	60	125	<sup>210</sup> Pb (Model)	Olefeldt et al. in prep
<b>SC3-</b> Е	61°18'N 121°18'W	29	60	<sup>210</sup> Pb (Model)	Olefeldt et al. in prep
SC3-YB	61°18'N 121°18'W	55	105	<sup>210</sup> Pb (Model)	Olefeldt et al. in prep
SC4-E	61°18'N 121°18'W	47	145	<sup>210</sup> Pb	Olefeldt et al. in prep
SC4-YB	61°18'N 121°18'W	62	120	<sup>210</sup> Pb	Olefeldt et al. in prep
L1-YB	59°28'N 117°10'W	64	130	<sup>210</sup> Pb (Model)	Heffernan et al. in prep
L1-IB	59°28'N 117°10'W	42	76	<sup>14</sup> C	Heffernan et al. in prep
Maximum age of transition			145		
Mature thermoka	rst bog cores				
SC1-OB	61°18'N 121°18'W	130	530	<sup>14</sup> C	Pelletier et al. 2017
SC2-OB	61°18'N 121°18'W	89	367	<sup>14</sup> C	Olefeldt et al. in prep
SC3-OB	61°18'N 121°18'W	66	143	<sup>14</sup> C	Olefeldt et al. in prep
SC4-OB	61°18'N 121°18'W	120	587	<sup>14</sup> C	Olefeldt et al. in prep
L1-OB	59°28'N 117°10'W	89	362	<sup>14</sup> C	Heffernan et al. in prep
Minimum age of transition			143		

Table S3.3. Summary statistics from peatland classification showing sites with site coordinates,	
number of 250 x 250m sections classified and percent coverage of the three peatlands classes $\pm$	
one standard deviation.	

Site	Site Coordinates	Date of Fire (size, ha)	Elevation (masl)	Classified sections (n)	Peat Plateau coverage (%)	Young Thermokarst Bog coverage (%)	Mature Thermokarst Bog coverage (%)
Zama Unburned	59°20'N	YN July 10, 9'W (6132)	584	55	74.3 ± 9.2	2.7 ± 1.3	23.0 ± 8.5
Zama Burned	119°19'W			37	66.8 ± 10.0	$6.2 \pm 2.8$	$26.7 \pm 9.1$
Trout Lake Unburned	60°26'N	June 28,	550	34	70.3 ± 13.3	2.7 ± 1.9	23.9 ± 12
Trout Lake Burned	120°51'W	(13,202)	559	57	65.52 ± 8.4	$6.3 \pm 3.3$	27.8 ± 8.2
Fort Simpson Unburned	62°25'N	August 3,	•	60	49.4 ± 10.7	5.1 ± 3.4	44.7 ± 11.8
Fort Simpson Burned	121°20'W	1°20'W 1995 (238,566)	260	43	52.9 ± 11.1	9.0 ± 3.9	37.7 ± 9.0
Sixtieth Burned	59°51'N	July 12,		61	$65.8 \pm 8.4$	13.0 ± 4.8	21.1 ± 8.0
Sixtieth Unburned	116°33'W	116°33'W (552,539)		58	55.0 ± 10.0	10.5 ± 3.9	34.3 ± 9.9

**Table S2.4.** Accuracy assessment for predictions of absence or presence of taliks within a burned peat plateau site using September thaw depths. Data is from one unburned site, Lutose Unburned (See table S2.1), where actual absence or presence of taliks was determined by seasonal monitoring of thaw depths.

	Predicted Absence of Talik	Predicted Presence Talik	of	Sum
Actual Absence of Talik	63	4		67
Actual Presence of Talik	4	24		28
Sum	67	28		95

**Overall accuracy = 92%** 

**Table S2.5.** Accuracy assessment for predictions of absence or presence of taliks within a burned peat plateau site using September thaw depths. Data is from one burned site, Lutose 2012 (See table S2.1), where actual absence or presence of taliks was determined by frequent monitoring of thaw depths.

	Predicted Absence of Talik	Predicted Presence Talik	of Sum
Actual Absence of Talik	36	2	38
Actual Presence of Talik	1	55	56
Sum	37	57	95

**Overall accuracy = 97%** 



**Figure S2.1.** Example of a peat plateau site (Kakisa 2014) where soil thermal regime was monitored. Each peat plateau site had a 5 m spaced grid with 100 point locations marked with flags where depth to frost table was measured.



**Fig. S2.2.** Example of transects from peat plateau, through young thermokarst bog, to mature thermokarst bogs. (a) Lutose 2012, b) Samba deh 2013, c) Fort Simpson unburned, d) Lutose unburned. Notice the shift in dominant vegetation between the young and mature thermokarst bogs.



**Fig S2.3.** Determination of talik presence, and estimation of thaw depth at point locations where taliks were absent. A) Example of seasonal development of the thawed soil layer at 6 representative point locations at the Lutose 2012 site. Soil probe was 150 cm long, hence measures of 150 cm indicate that no frost table was reached. We classified point locations as having taliks if there was a >90 cm increase in thaw depth between two monitoring occasions, as exemplified by three point locations. Only a subset of sites had thaw depths monitored as frequently as Lutose 2012, and thus for other sites we did not have direct observations of talik frequency. B) Histogram of September point location depth to frost table measurements (n = 100) from the Fort Simpson 1995 site overlaid by a gaussian kernel density estimation (Wessa 2015). All point locations with September thaw depths >100 cm were assumed to have taliks, an assumption that was tested against observed talik presence at a few sites (See table S2 and S3). We used the maximum density to estimate representative active layer depth of point locations where taliks were absent, as using either the mean or median thaw depths would be strongly influenced by talik frequency.



**Fig S2.4**. September active layer depth with mean annual air temperature in the burned and unburned sites. Climate was not found to be the main influence on active layer depth, supporting our conclusion that wildfire is the main driver of difference in active layer depth.



**Fig S2.5.** Differences in the seasonal development of the thawed soil layer between unburned and historically burned sites. A) Thaw depths are shown  $\pm 1$  standard deviation of n=3 (2-5 years), n=4 (7-22 years), and n=3 (32-55 years). Point locations where talks later developed were included for June and July dates. B) Absolute difference in depth to frost table between burned and paired unburned sites in early June. Sites 10, 21, 32 post fire not included due to inaccessibility at this time of year.



**Fig S2.6.** Example section of World View 2 satellite image. Supervised classification was carried out for 250 x 250 m sections of peatlands that were visually assessed to only contain peat plateau, young thermokarst bog, and mature thermokarst bogs. Sections containing channel fens, upland forests, ponds, clouds, or shadows were excluded from analysis and are shown with grey shading.

**Burned Sections** 



Fig S2.7. Field validation of extents of young thermokarst bog. World View 2 Satellite Image  $(75 \times 75 \text{ m})$  shown right with corresponding classification on the left. Overlain points taken with dGPS unit, up to 10 cm accuracy.



**Fig. S2.8.** Residual difference between edges of supervised classification and field measurements of edge. Overestimates occurred when classification is outside the field delineated young thermokarst bog area, underestimates occurred when the classification was within the field delineated young thermokarst bog area.



**Figure S2.9.** Estimate of the trajectories of young thermokarst bog development. A) Trajectories of recent thermokarst bog development based on assumptions 1-4. B) 95% confidence intervals for each site, from which mean and standard deviation of the mean were used for final trajectory estimate.

## 2.9.10 Supplementary Literature Cited:

- 1. Hugelius G, et al. (2013) The Northern Circumpolar Soil Carbon Database: spatially distributed datasets of soil coverage and soil carbon storage in the northern permafrost regions. *Earth Syst Sci Data* 5:3–13.
- 2. Canada NR (2017) Canadian Wildland Fire Information System. Available at: http://cwfis.cfs.nrcan.gc.ca/datamart/metadata/nfdbpoly [Accessed July 14, 2017].
- 3. Scott D. Kernal Density Estimators in Multivariate Density Estimation: Theory, Practice, and Visualization (John Wiley & Songs Inc, Hobken, NJ, USA).
- 4. Wessa P (2017) Free Statistics Software, Office for Research Devlopment and Education, version 1.1.23-r7. Available at: http://www.wessa.net/.
- 5. Wright N, Hayashi M, Quinton WL (2009) Spatial and temporal variations in active layer thawing and their implication on runoff generation in peat-covered permafrost terrain. *Water Resour Res* 45(5):1–13.
- 6. Jonasson G (1983) The point intercept method for non-destructive estimation of biomass. *Phytocoenologia* 11(3):385–388.
- 7. Baltzer JL, Veness T, Chasmer LE, Sniderhan AE, Quinton WL (2014) Forests on thawing permafrost: Fragmentation, edge effects, and net forest loss. *Glob Chang Biol* 20(3):824–834.
- 8. Mamet SD, Chun KP, Kershaw GGL, Loranty MM, Kershaw GP (2017) Recent Increases in Permafrost Thaw Rates and Areal Loss of Palsas in the Western Northwest Territories, Canada. *Permafr Periglac Process*. doi:10.1002/ppp.1951.
- 9. Chasmer LE, Hopkinson C (2016) Threshold loss of discontinuous permafrost and landscape evolution. *Global Biogeochem Cycles* doi: 10.11:1–15.

# <u>Chapter 3: Effects of wildfire on soil respiration in permafrost peatlands of the Canadian</u> <u>Western Boreal Forest</u>

#### **3.1 Abstract**

Wildfire in northern boreal peatlands is contributing to widespread permafrost thaw by deepening the active layer, causing significant amounts of previously frozen soil carbon to be exposed to microbial decomposition. This may result in the release of old carbon sources to the atmosphere in the form of the greenhouse gas  $CO_2$  but the rate and duration of such release is poorly quantified. We used field-based measurements of soil respiration in combination with a two-depth soil temperature model in burned and unburned sites to determine carbon loss due to fire-induced active layer deepening. We show that respiration during the growing season is significantly lower in burned sites, despite having active layers over 100cm deeper than unburned site. However, we estimate the proportion of carbon respired from depth to be 40% in the burned site, which includes previously frozen carbon stores, compared to 10% in the unburned site. This results in ~32 g Cm<sup>-2</sup> respired from depth in the burned site and ~8 g Cm<sup>-2</sup> was loss from the unburned peatlands, likely due to the removal of the top labile peat layer during fire, fire-induced active layer deepening mobilizes carbon release from these ecosystems.

#### **3.2 Introduction**

Northern boreal forests in the discontinuous permafrost zone are experiencing drastic warming and increased fire frequencies [*Kasischke and Turetsky*, 2006] that is expected to alter the carbon cycling in these regions [*Schuur et al.*, 2015]. These altered conditions are likely to result in changes in soil temperatures and permafrost stability [*Genet et al.*, 2013; *Zhang et al.*, 2015; *Helbig et al.*, 2016] that control greenhouse gas fluxes to the atmosphere. Northern permafrost soils hold large quantities of carbon [*Hugelius et al.*, 2014] that is considered locked as it undergoes minimal microbial activity while frozen. However, deeper seasonal thaw layers as the result of permafrost thaw [*Viereck et al.*, 2008; *Brown et al.*, 2015] and thermokarst formation [*Olefeldt et al.*, 2016] expose previously frozen soil organic carbon to decomposition, potentially incorporating it into current carbon cycling. The magnitude, timing, and form of the

permafrost carbon feedback remains highly uncertain [*Schuur et al.*, 2015] and increasing prevalence of disturbances such a wildfire on the landscape are furthering our uncertainty about potentially large positive feedbacks from these systems [*Schaefer et al.*, 2012; *Treat et al.*, 2013; *Koven et al.*, 2015].

Wildfire is the major stand-renewing agent in these systems and is an integral part in determining patterns of vegetation and forest structure [*Rowe and Scotter*, 1973]. Wildfire within peatlands represent a large source of carbon to atmosphere, burning over 1500km<sup>2</sup> and releasing 6TgC annually in Western Canada alone. Wildfire affects forested permafrost peat plateaus in similar frequencies to upland forests (100-120 year fire return interval; [*Turetsky et al.*, 2004]) and in the last 30 years more than 25% of permafrost peatlands within the Taiga Plains ecozone have burned [*Gibson et al. in prep*]. Climate change is likely to exacerbate the effect of fire on carbon emissions from peatlands due to changes in the fire regime and drought frequency and intensity [*Flannigan et al.*, 2009]. Across North America the total annual burn area has increased due to higher mean annual temperatures and drier conditions [*Kasischke and Turetsky*, 2006] and is predicted to increase by 25% by the year 2030 and the 75-100% by the end of the century [*Wotton et al.*, 2010].

The increase in fire prevalence in these permafrost landscapes is having significant impacts on permafrost stability, and therefore the stability of these soil organic carbon reserves. Studies have shown that fire can increase soil temperatures for several decades and lead to thickening of the active layer [*Mackay*, 1995; *Burn*, 1998; *Yoshikawa et al.*, 2002; *Viereck et al.*, 2008; *Rocha et al.*, 2012; *Nossov et al.*, 2013; *Brown et al.*, 2015, *Gibson et al. in prep*]. Despite the rapid recovery of understory vegetation that has been shown to dramatically increase surface albedo [*Randerson et al.*, 2006; *Amiro et al.*, 2009], the most pronounced effect of wildfire on soil temperatures in permafrost peatlands occurs 10-20 years post fire where soils temperatures at depth can be as much as 4°C warmer [*Gibson et al. in prep.*]. Many of the observations on active layer thickening come from Alaskan upland systems or lowlands with shallow peat layers (<1m [*Viereck et al.*, 2013]. However, in the permafrost peatlands of Western Canada where peat depths range from 2 to 6 m, active layers can increase up to 100cm in depth after a fire [*Gibson et al. in prep.*]. This combination of increased soil temperatures and increased seasonally thawed

soils would tend towards an increase in net carbon loss from post-fire permafrost soil. However, we have very few measurements on this phenomenon.

Much of our current understanding of the response of soil respiration from thawed permafrost soils comes from incubations experiments. Soil microbial respiration generally increases exponentially with temperature [*Davidson and Janssens*, 2006] and Schadel et al. (2016), using meta-analysis from 21 incubation studies, showed that increasing soil temperatures led to a net increase in carbon release by a factor of 2.0. Furthermore, the amount of carbon released under aerobic conditions was on average 3.4 times higher than anaerobic incubations conditions. This is particularly important in post fire peat plateau systems where permafrost thaw leads to a deepening and warming of the aerobic layer, which could drive the mobilization of stored older carbon.

The temperature dependence of soil respiration may be modeled by a constant  $Q_{10}$  which is the proportional increase in the rate of respiration for a change of 10°C and is well established to range between 2 and 2.5 [eg. *Knorr et al.*, 2005]. Most models use near surface temperatures which are found to be relatively good predictors of ecosystem respiration [*Lafleur et al.*, 2005]. This is likely because a large portion of CO<sub>2</sub> released comes from near surface sources associated with vegetation and the top soil layer which experiences the largest variation in temperature and represents the greatest fraction of labile carbon in the soil. However, such near-surface temperature models do not seem suitable to account for the respiration at depth when the soil thermal regime at depth is altered, particularly as a consequence of fire-induced active layer deepening. To our knowledge no study has used two temperatures (both a shallow and deep temperature, or a temperature profile) to model soil respiration within permafrost landscapes. This method would allow us to isolate contribution of soil respiration from shallow and deep old carbon sources. Given there is generally a higher temperature sensitivity of soil respiration at low temperatures [*Kirschbaum*, 1995; *Goulden*, 1998], soil organic carbon in cold regions may be highly sensitive to above-average temperatures that occur as a result of wildfire.

Previous work by Schuur et al. (2009) found that following permafrost thaw with thermokarst development up to 15% of ecosystem respiration is derived from the decomposition of old carbon. Yet presently, there remains little understanding of contribution from deep, old carbon to soil respiration following active layer deepening, particularly following a wildfire where soil temperatures at depth have been shown to be much greater. In order to address this we

use *in situ* measurements to examine the effect of wildfire on post fire soil respiration from sites that burned 10-20 years ago and a paired unburn site to (1) quantify and compare soil respiration from burned and unburned permafrost peatlands, (2) determine if soil respiration models that include both a shallow and a deep temperature more accurately determine carbon loss from post fire sites, (3) and examine the contribution of soil respiration from deep, fire-induced thawed soils to total soil respiration within a site.

#### 3.3 Methods

## 3.3.1 Study Area

We conducted our study within a burned and paired peatlands near Lutose, AB (59°29'4N, 117°10'43W) within the discontinuous permafrost zone of the Taiga Plains (Figure 3.1). We selected sites that burned in 2000 and 2007 and a paired unburned site (Figure 3.1). Peatlands in this area are a mosaic of permafrost plateaus, channel fens, and ombrotrophic flat bogs. Unlike surrounding bogs and fens, the permafrost peat plateaus support black spruce (Picea mariana) and are underlain by ~5-10m thick permafrost. The mean annual air temperature during our study period was -1.5°C. Vegetation within the burn sites is limited to *Rhododendron groenlandicum* and <1m tall black spruce while the unburned site is dominated by black spruce  $2.3 \pm 1.9$ m tall and an understory of predominantly *Rhododendron* groenlandicum, Vaccinimum vitis-idaea, and Vaccinium uliginosum, Cladonia spp. and Sphagnum spp. Both burn sites show little to no nonvascular recovery with 95-100% char cover compared to the unburn site that is 80% lichen (Cladonia rangiferina). Both burn sites had thaw depths >150cm by the end of the summer compared to the unburn site which was ~52cm. The burned site displayed similar pre fire stand density as the unburned sites and is within three kilometers of both sites, as such we do not believe the unburned site differed significantly from the burn sites pre fire.

## 3.3.2 Field data collection

For all our measurements we selected flat areas dominated by lichen (unburned site) or char (burned sites) surfaces, avoiding hummocks and hollows as these micro-topographical features are known to alter soil thermal regimes [*Wright et al.*, 2009]. These flat areas are

representative of the studied sites where hummocks covered 5% (2000 burned site), 14% (2007 burned site) and 6% (unburned site) and hollows are mostly absent (burned sites) or covered 7% (unburned site) of the plateau area.

Soil temperature loggers (Pendant HoboProV2, Onset Corp.) were installed at a depth of 10cm and 40cm in two locations in each site during summer 2015. Air temperature was measured using a temperature logger at 1m above the ground, shielded from direct sunlight. Soil temperature time-series were obtained by recording measurements at two-hour intervals. Temperatures were averaged between the two loggers within each site due to periodic sensor failure, which occurred during the month of July in the 2000 burn site. Air and soil temperatures were measured from May 2016 to July 2017. Soil moisture sensors (Hobo Smart Sensor 10HS) were installed in July 2016 at 25 and 35 cm and recorded soil moisture every 20 minutes until July 2017.

Soil respiration fluxes were measured biweekly during the summer (Julian day 150 - 300) using static chambers. In each site, six collars (diameter = 0.4m) were installed 10cm into the soil. While our soil respiration measurements account for heterotrophic and autotrophic sources we consider the autotrophic component to be low due to the absence of vascular vegetation in our selected locations and the trenching of shallow roots during collar installation. Flux measurements started two weeks after collar installation using PVC chambers (volume = 37.7L) sealed with adhesive tape to the collars. A thermometer inside the chamber measured temperature in the headspace and carbon dioxide concentrations were recorded every 1.6 s for 2-3 minutes using a PP Systems EGM-4 portable infrared gas analyzer (IRGA; Amesbury, Massachusetts). The CO<sub>2</sub> flux rate was calculated as the slope of the linear relationship between headspace  $CO_2$  concentrations and time. We excluded any slopes with  $r^2$  less than 0.8, which represented less than 3% of our measurements. Soil temperature at 5, 10, and 40cm depth was recorded using a handheld thermometer during CO<sub>2</sub> flux measurements. We used soil temperatures from the loggers at 40cm when thermometers were not available (May). A comparison between thermometers and logger data at 10cm and 40cm showed no significant difference between both methods (t=0.9235, p<0.05). Soil moisture was also measured at the soil surface in four locations adjacent to the collar using a Delta-T HH2 portable soil moisture meter. Soil moisture was examined as a possible control on respiration. No relationship between soil moisture and soil respiration was found in either the burned or the unburned site (Figure

S2.1). Therefore, soil moisture was not a controlling factor and thus only soil temperature was used in soil respiration modelling. Several studies have shown both soil temperature and soil moisture to be an important control on soil respiration. However, there are controversial findings about the relative importance of soil moisture in controlling soil respiration in peatlands, with the relative importance of it in drier peatlands like peat plateaus being unknown [*Lafleur et al.*, 2005]. Although Natali et al. (2015) showed a positive correlation between monthly soil respiration and water table depth in a tundra upland, in peat plateaus water content at depth is relatively constant (Figure S3.1) and in general  $CO_2$  production is most sensitive to moisture changes in the uppermost portions of the peat profile [*Lafleur et al.*, 2005]. In our study soil moisture content was relatively stable in the upper layers, thus the little change in heterotrophic respiration with soil moisture is not unlikely.

## 3.3.3 Modeling and statistical analysis

We examined the relationship between respiration and soil temperature using regression analysis. Soil moisture was not found to be a predictor of respiration (Figure S3.1) and was excluded from the analysis. Data from both burned sites was combined to evaluate the relation between temperature and soil respiration. Soil temperature and soil moisture did not differ between the two burned sites (p > 0.05, Figure 3.2a and b, paired t-test), rendering confidence in combining data points from the two burned sites. Soil temperature from the six collars was averaged for each site in during each measurement day for the analysis. We used repeated measures ANOVAs to determine differences between soil respiration in burned and unburned sites.

We examined the temperature dependence of ecosystem respiration using nonlinear least square regression during the growing season using the following one- and two-depth soil temperature models:

(1) 
$$R = (a * b^{(T5/10)}*BURN) + (c * b^{(T5/10)}*UNBURN)$$
  
(2)  $R = (a * b^{(T10/10)}*BURN) + (c * b^{(T10/10)}*UNBURN)$   
(3)  $R = (a * b^{(T40/10)}*BURN) + (c * b^{(T40/10)}*UNBURN)$   
(4)  $R = [(a * b^{(T5/10)} + c * b^{(T40/10)})*BURN] + [(d * b^{(T5/10)} + e * b^{(T40/10)})*UNBURN]$   
(5)  $R = [(a * b^{(T10/10)} + c * b^{(T40/10)})*BURN] + [(d * Q_{10}^{(T10/10)} + e * b^{(T40/10)})*UNBURN]$
where R is the rate ecosystem respiration ( $\mu$ mol C m<sup>-2</sup> s<sup>-1</sup>), a,c,d,e are fitted parameters from the regression representing the rates of ecosystem respiration at 0°C ( $\mu$ mol C m<sup>-2</sup> s<sup>-1</sup>) from the associated temperature depth, b is the  $Q_{10}$  - the temperature dependence of ecosystem respiration, T is the soil temperature at either 5, 10, or 40cm (°C) beneath the soil surface, and BURN/UNBURN are dummy variables of either 1 or 0 depending on the site the model is acting upon. The a, b, c, d, and e parameters were defined by using nonlinear lead squares (nls) package in R [*Bates and Chambers, 1992*].

The Akaike information criterion (AIC) was used to select the best model for both the burned and the unburned sites over the use of an  $R^2$  due to its likelihood to exhibit an extreme bias to highly parameterized non-linear models [*Spiess and Neumeyer*, 2010]. The contribution of deep respiration during the growing season was calculated using the b \*  $Q_{10}^{(T40/10)}$  component of the model and dividing that by the total respiration (R). We compared the contribution of shallow (10cm) and deep (40cm) soil to soil respiration between sites the most parsimonious model and the soil profile temperatures measured near the collars during flux measurements.

Estimates of the cumulative soil respiration during the growing season at each site was calculated with the best fit model using soil temperature time-series from the loggers (Figure S3.2). Soil temperature was averaged from two loggers in the unburned site and from four loggers in the two burned sites. Model coefficients plus standard error was used for statistical bootstrapping with 300 iterations to determine an instantaneous flux rate for each 2-hour interval. These rates were then used to calculate cumulative respiration over the growing season for each site (JD 181 - 259).

# 3.4 Results

#### 3.4.1 Soil Temperature

Burned sites had deeper active layers and warmer soils than the unburned site. During the summer period (Julian Day 150 – 250), soil temperature at 10 and 40cm was significantly higher in the burned sites than in the unburned site (p<0.05, paired t-test), although no differences between sites were observed nearer to the surface at 5 cm. Burned soils were over 2°C warmer,

and over 4°C warmer at 10 and 40 cm respectively in the burned sites mid-summer (Figure 3.2a and Figure 3.2b).

Soil temperatures decreased with depth during the growing season and the difference between depths became smaller towards late fall, when the gradient with depth ended and temperatures at 10 and 40 cm converged and remained similar during the winter months. This occurred with a delay of about 25 days in the burned sites in comparison to the unburned site.

# 3.4.2 Soil respiration

Soil respiration had a seasonal response peaking towards the middle of the growing season in all sites. Rates were significantly higher in the unburned site than in both burned sites during the growing season (F=11.42, p<0.05, Figure 3.3a) and no significant differences occurred between the two burned sites (F=0.025, p>0.05, Figure 3.3a). Fluxes in the burned sites increased from (average ±1 standard deviation)  $0.60\pm0.28 \ \mu\text{mol} \ \text{C} \ \text{m}^{-2} \ \text{s}^{-1}$  in early summer to  $0.90\pm0.21 \ \mu\text{mol} \ \text{C} \ \text{m}^{-2} \ \text{s}^{-1}$  in the unburned site, respectively. Overall, soil respiration was 27% greater on average in the unburned site than in the burned sites during the growing season period and the maximum difference occurred the early summer, at 46%. Soil temperature was exponentially related to soil respiration in both the burned and unburned site (Figure 3.3b) supporting the use of an exponential equation to model soil respiration.

# 3.4.3 Soil respiration models

The results of the regression models between respiration and temperature are presented in Table 1 and 2. The model that includes soil temperatures at both 10 and 40cm depth (model 1) was selected as the best fit based on its AIC value (Table 3.1 and 3.2). Incorporating the 40 cm soil temperature improved the model performance (compare model 1 with 2 and model 3 with 4) with the model containing only the 40 cm temperature performing the poorest (model 5). The models including the soil temperature at 10 cm fitted better to the data (model 1 and 2). The temperature dependence of soil respiration on temperature ( $Q_{10}$ ) obtained from the modelled *b* parameter was 2.56±0.3. The most parsimonious model (model 1), which included, two temperatures was able to accurately model soil respiration during the growing season (Figure 3.4).

### *3.4.4 Deep peat contribution to CO*<sub>2</sub> *release*

The burned sites had nearly 4 times as much respiration from depth (40cm) where soils thawed two weeks earlier than in the unburned site.

The proportion of soil respiration coming from deep layers in the burned sites increased 15% during the growing season from,  $\sim$ 36% in the early summer to  $\sim$ 55% in the late fall. This is, contrast to the unburned site that only experienced a 5% increase in respiration during the growing season, from  $\sim$ 9% in the early summer to  $\sim$ 14% in the late fall. (Figure 3.5).

We calculated the cumulative carbon loss from shallow and deep soil in the burned and unburned sites (Figure 3.6). While total CO<sub>2</sub> release was greater in the unburned site compared to the burned site during the growing season  $83.92 \pm 1.79$  gC m<sup>-2</sup> vs  $76.58 \pm 1.25$  gC m<sup>-2</sup> during the growing season (JD 150-258), deep peat sources (40cm) contributed substantially in the burned sites with a cumulative loss of  $32.11 \pm 0.97$  gC m<sup>-2</sup> compared to  $8.17 \pm 0.96$  gC m<sup>-2s</sup> in the unburned site (Figure 3.6b). By the end of the season this represents nearly 4 times as much carbon loss from deep soils to the atmosphere in burned sites.

## **3.5 Discussion**

Recent research has shown that wildfire within the discontinuous permafrost zone can cause a deepening of the active layer in boreal permafrost peat plateaus [*Viereck et al.*, 2008; *Brown et al.*, 2015] which exposes large expanses of previously frozen soil organic carbon to microbial decomposition. Furthermore, not only do active layers deepen post-fire, deep (40cm) soil temperatures can rise as much as 4°C within permafrost peat plateaus [*Gibson et al. in prep*] which has been shown to stimulate soil respiration [*Davidson and Janssens*, 2006]. Therefore, in this study, our objective was to determine if fire induced deepening of the active layer leads to increased total soil respiration compared to unburned sites, and if it stimulates the respiration of old, previously frozen soil organic carbon. In order to do this, we explored the use of a two-temperature (shallow and deep temperature) soil respiration model, as a tool that may more accurately model soil respiration and allow us to partition respiration into both shallow and deep components.

Our first objective aimed to compare soil respiration from both a burned and unburned permafrost peat plateau. We hypothesized that, due to warmer temperatures and deeper active

layers, the burned sites would have a higher rate of respiration compared to the unburned site. Surprisingly, the unburned site was found to have the higher rate of soil respiration at approximately  $0.83\pm0.44$  µmol C m<sup>-2</sup> s<sup>-1</sup>, compared to  $0.66\pm0.30$  µmol C m<sup>-2</sup> s<sup>-1</sup> in the unburned site, averaged over the course of the growing season (JD150-300). We found that the unburned site had consistently higher overall rates of respiration than the burned sites until mid-fall when rates of respiration were near equal and a third of what they were during peak growing season. We attribute higher soil respiration rates in the unburned site to the source and quality of carbon that is being respired in the burned versus the unburned site. During a fire event, the top 5-20cm of peat is combusted [Turetsky et al., 2011]. This peat has recent inputs of fresh, labile carbon that exhibits higher turnover rates, as opposed to carbon at depth that is more recalcitrant [Bosatta and Ågren, 1999; Knorr et al., 2005]. Therefore, even a low severity fire that removes just 5 cm of organic soil is likely to result in the loss of the labile carbon pool. We attribute the observed lower respiration rates of the burned sites to soil microbes that favour these easily decomposable organic compounds found in the top layers. This results in the high rates of decomposition near the surface, and decreasing rates with depths as carbon pools become less labile [Bergeron, 2000]. Interestingly, despite our findings that post fire increases in soil temperatures do increase respiration depth, we conclude that this is not enough to overcome the much higher respiration rates of labile surface peat. As a result of this, we conclude that wildfire does not cause these ecosystems to become larger sources of carbon to the atmosphere.

Our results are contrary to that of Dorrepaal et al. (2009), who showed long-lasting sensitivities to climate warming by soil carbon towards the bottom of the active layer in subarctic peatlands. They found the fresh surface litter and plant-related respiration contributed proportionally less to total respiration than in the most other ecosystems because of the large, organic soil pool at depth. Our results also support the conclusion of increased respiration at depth due to increased soil temperatures though they do not support the notion the fresh surface litter inputs are not as significant in total respiration (Figure 6). Despite burned sites having a large proportion of total respiration coming from depth, the higher yearly respiration from the unburned site that has over 100cm less thawed organic soils highlights the importance of these fresh surface litter inputs.

Despite our results being limited to the growing season, we observed a trend towards increasing respiration from the burned sites towards the end of the season compared to the

unburned site that displayed as faster rate in decline (Figure 3). During this time period (JD>280) we observed convergence of the surface temperature, but the temperatures at depth remained warmer in the burned site. We suspect that as a consequence of this, respiration remains higher at depth in the burned site during a period of time where surface respiration is relatively minimal. Although outside the scope of this study, if this trend were to continue during the winter months it could have a noticeable effect on the net carbon loss from these soils on an annual basis. As well, during this time period there is decreased fresh litter inputs as vegetation begins to senesce, likely contributing to decreased respiration at the surface in unburned sites.

Our second objective in this study was to examine models for soil respiration that used one or two temperatures. By using a two-temperature model we were able to effectively partition respiration into shallow (10cm) and deep (40cm) respiration. Most attempts at partitioning respiration have come from incubation studies (see [*Schädel et al.*, 2016] ) or  $\Delta$ 14C measurements (see [*Dutta et al.*, 2006; *Hilasvuori et al.*, 2013]). To the best of our knowledge, our study is one of the first to model soil respiration with two temperatures. In general, a single temperature at a shallow depth is used to model soil respiration [e.g. *Chivers et al.*, 2009]. This is often because single temperature models that use only deep temperatures poorly represent the data. This was also observed in our study with the single temperature at 40cm being our worst performing model based on an AIC evaluation (Table 1). However, we show that when used in conjunction with a shallow temperature not only does a two temperature model effectively model the data (Figure 5) it can be used as a new way to model respiration from shallow and deep portions of the soil profile.

Our modeled temperature sensitivity of decomposition ( $Q_{10}$ ) was similar to those seen in other studies that incubated active layer soils, 1.8 - 2.5. Here we applied the same  $Q_{10}$  value for both our shallow and deep soil contributions, though recognize that these two depths may have different sensitives as deeper, more recalcitrant carbon generally has a higher temperature sensitivity [*Dorrepaal et al.*, 2009]. However, due to the constraints of our models, it was necessary to set them as equivalent. Other studies that have examined temperature sensitives on peat plateaus have reported significantly higher  $Q_{10}$  values (example, [*Startsev et al.*, 2016], which reported 5.6 - 11.5) though this is not generally accepted as biologically valid, and therefore may be an artifact of data sampling and analysis. There are several environmental constraints that can affect the temperature sensitivity of organic material to decomposition. While freezing severely limits microbial activity, upon thaw both chemical and biological factors can limit the temperature sensitivity of decomposition. Many incubations studies have shown a sharp decline in soil respiration rates within the first 3 weeks of the incubation, and an increase in the temperature sensitivity and respiration rates due to the depletion of the labile carbon pool [*Conant et al.*, 2008; *Xu et al.*, 2010]. This idea supports our findings that temperature sensitivities and respiration in a burned site would be lower due to the loss of the labile carbon pool during the fire event.

Our last objective was to derive estimates of deep respiration to quantify the release of thawed carbon in burned sites and assess the effects of disturbance on the carbon cycle in permafrost peatlands. Our finding that wildfire increased respiration from depth is consistent with the work of [*Dorrepaal et al.*, 2009] that showed that at least 69% of an observed increase in respiration with warming originated from carbon in deeper peat layers. In our study we show that up to 55% originates from deeper peat layers following a fire event, likely due to upwards of 100cm deeper active layers in the burned than unburned site, suggesting the mobilization of old carbon from depth. Collectively, warming of soils at depth, whether via climate warming or wildfire, is accelerating the respiration of subsurface carbon reservoirs to a much larger extent than previously thought [*Christensen et al.*, 1999; *Hogg et al.*, 1999].

Surprisingly though, we found that the unburned site had consistently higher overall rates of respiration than the burned sites. As discussed earlier, we attribute this to the source and quality of carbon that is being respired. As a result of this labile carbon pool we conclude that the increased respiration from depth isn't enough to overcome the importance of this labile pool. Regardless, wildfire does result in old carbon being incorporated into the current carbon cycle and given present day climate warming and permafrost thaw rates it is unlikely that this newly incorporate pool of carbon will be fixed and preserved in permafrost once again, representing a long term source of carbon in our atmosphere.

# **3.6** Conclusion

As fire frequencies and severity changes in northern boreal forests as a result of climate change, widespread permafrost thaw is likely to occur, and cause drastic changes to carbon cycling in these environments [*Schuur et al.*, 2015]. Our study shoes that when permafrost peat

plateaus undergo fire, and subsequent active layer deepening [*Gibson et al. in prep*], they do not become larger sources of carbon to the atmosphere in the years to decades following the fire event due to increased soil respiration. This is despite having over 100cm more thawed soils and 4°C warmer soils at depth. Our study disputes that contribution of soil respiration, from exposed, deep active layers, will be a greater source of carbon to the atmosphere than fresh surface litter.

This study revealed that despite higher temperature sensitivities at depth, burned permafrost peat plateaus actually had lower soil respiration than unburned peat plateaus. Furthermore, this occurred despite significantly higher proportion of respiration from depth in the burned site, highlighting the likely importance of fresh surface litter inputs that are burned off during the fire. These findings have important implications for prediction of future climate changes in permafrost peatland ecosystems. It is important to reconcile the discrepancy between modeled and observed increases in soil respiration because models that account for increased temperature sensitivities at depth, but neglect the loss of labile surface carbon during burning, will likely overestimate post fire soil respiration in permafrost peat plateaus. Ultimately, wildfire is leading to the loss of old soil carbon from depth that was previously stored in the terrestrial environment. As the extent of wildfire increases across the discontinuous permafrost zone, the rate of old carbon mobilization to the atmosphere will increase. What our results also highlight is the potential large soil carbon loss that could exists in the absence of fire, but in the presence of climate warming. If permafrost peat plateaus experience deepening of the seasonal active layer but also maintain the fresh sources of labile carbon inputs larger than expected rates of soil respiration could occur as carbon is mobilized from depth.

## 3.7 Acknowledgements

We thank Liam Heffernan, Katheryn Burd, McKenzie Kuhn, Michael Barbeau and Jessi Steinke for assistance with field work. We greatly acknowledge financial support for the research from the following sources: National Science and Engineering Research Council Discovery grant (RGPIN-2016-04688), the Campus Alberta Innovates Program, the University of Alberta Northern Research Awards, and the Northern Scientific Training Program.

#### 3.8 References:

- Amiro, B. D., a. Cantin, M. D. Flannigan, and W. J. de Groot (2009), Future emissions from Canadian boreal forest fires, *Can. J. For. Res.*, 39(2), 383–395, doi:10.1139/X08-154.
- Bates, D. M. and Chambers, J. M. (1992). Nonlinear models. In Chambers, J. M. and Hastie, T. J., editors, Statistical Models in S, pages 421–454. Wadsworth, Pacific Grove, CA.
- Bergeron, Y. (2000), Species and Stand Dynamics in the Mixed Woods of Quebec's Southern Boreal Forest, *Ecology*, *81*(6), 1500–1516.
- Bosatta, E., and G. I. Ågren (1999), Soil organic matter quality interpreted thermodynamically, *Soil Biol. Biochem.*, *31*(13), 1889–1891, doi:10.1016/S0038-0717(99)00105-4.
- Brown, D. R. N., M. T. Jorgenson, T. A. Douglas, V. E. Romanovsky, K. Kielland, C. Hiemstra, E. S. Euskirchen, and R. W. Ruess (2015), Interactive effects of wildfire and climate on permafrost degradation in Alaskan lowland forests, *J. Geophys. Res.*, 120, 1619–1637, doi:10.1002/2015JG003033.Received.
- Burn, C. R. (1998), The active layer: Two contrasting definitions, *Permafr. Periglac. Process.*, 9(4), 411–416, doi:10.1002/(SICI)1099-1530(199810/12)9:4<411::AID-PPP292>3.0.CO;2-6.
- Chivers, M. R., M. R. Turetsky, J. M. Waddington, J. W. Harden, and a. D. McGuire (2009), Effects of experimental water table and temperature manipulations on ecosystem CO2 fluxes in an Alaskan rich fen, *Ecosystems*, 12(8), 1329–1342, doi:10.1007/s10021-009-9292-y.
- Christensen, T. R., S. Jonasson, T. V. Callaghan, and M. Havström (1999), On the potential CO2 release from tundra soils in a changing climate, *Appl. Soil Ecol.*, *11*(2–3), 127–134, doi:10.1016/S0929-1393(98)00146-2.
- Conant, R. T., R. A. Drijber, M. L. Haddix, W. J. Parton, E. A. Paul, A. F. Plante, J. Six, and M.

J. Steinweg (2008), Sensitivity of organic matter decomposition to warming varies with its quality, *Glob. Chang. Biol.*, *14*(4), 868–877, doi:10.1111/j.1365-2486.2008.01541.x.

- Davidson, E. a., and I. a. Janssens (2006), Temperature sensitivity of soil carbon decomposition and feedbacks to climate change, *Nature*, *440*(7081), 165–173, doi:10.1038/nature04514.
- Dorrepaal, E., S. Toet, R. S. P. van Logtestijn, E. Swart, M. J. van de Weg, T. V. Callaghan, and R. Aerts (2009), Carbon respiration from subsurface peat accelerated by climate warming in the subarctic, *Nature*, 460(7255), 616–619, doi:10.1038/nature08216.
- Dutta, K., E. A. G. Schuur, J. C. Neff, and S. A. Zimov (2006), Potential carbon release from permafrost soils of Northeastern Siberia, *Glob. Chang. Biol.*, 12(12), 2336–2351, doi:10.1111/j.1365-2486.2006.01259.x.
- Flannigan, M., B. Stocks, M. Turetsky, and M. Wotton (2009), Impacts of climate change on fire activity and fire management in the circumboreal forest, *Glob. Chang. Biol.*, 15(3), 549– 560, doi:10.1111/j.1365-2486.2008.01660.x.
- Genet, H. et al. (2013), Modeling the effects of fire severity and climate warming on active layer thickness and soil carbon storage of black spruce forests across the landscape in interior Alaska, *Environ. Res. Lett.*, *8*, 1–13, doi:10.1088/1748-9326/8/4/045016.
- Goulden, M. L. (1998), Sensitivity of Boreal Forest Carbon Balance to Soil Thaw, *Science (80-. ).*, *279*(5348), 214–217, doi:10.1126/science.279.5348.214.
- Helbig, M., K. Wischnewski, N. Kljun, and E. Laura (2016), Regional atmospheric cooling and wetting effect of permafrost thaw-induced boreal forest loss, *Glob. Chang. Biol.*, 22, 4048– 4066, doi:10.1111/gcb.13348.
- Hilasvuori, E., A. Akujärvi, H. Fritze, K. Karhu, R. Laiho, P. Mäkiranta, M. Oinonen, V. Palonen, P. Vanhala, and J. Liski (2013), Temperature sensitivity of decomposition in a peat

profile, Soil Biol. Biochem., 67, 47-54, doi:10.1016/j.soilbio.2013.08.009.

- Hogg, E., V. J. Lieffers, and R. W. Wein (1999), Potential Carbon Losses From Peat Profiles : Effects of Temperature , Drought Cycles , and Fire, *Ecol. Appl.*, 2(3), 298–306.
- Hugelius, G. et al. (2014), Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps, *Biogeosciences*, *11*(23), 6573–6593, doi:10.5194/bg-11-6573-2014.
- Kasischke, E. S., and M. R. Turetsky (2006), Recent changes in the fire regime across the North American boreal region—Spatial and temporal patterns of burning across Canada and Alaska, *Geophys. Res. Lett.*, 33(9), L09703, doi:10.1029/2006GL025677.
- Kirschbaum, M. U. F. (1995), the Temperature Dependence of Soil Organic Matter Decomposition, and the Effect of Global Warming on Soil Organic C Storage, *Soil Biol. Biochem.*, 27(6), 753–760.
- Knorr, W., I. Prentice, J. House, and E. Holland (2005), Long-term sensitivity of soil carbon turnover to warming, *Nature*, 433(January), 298–301, doi:10.129/2002PA000837.
- Koven, C. D., D. M. Lawrence, and W. J. Riley (2015), Permafrost carbon climate feedback is sensitive to deep soil carbon decomposability but not deep soil nitrogen dynamics, *Proc. Natl. Acad. Sci.*, *112*(12), 3752–3757, doi:10.1073/pnas.1415123112.
- Lafleur, P. M., R. a. Hember, S. W. Admiral, and N. T. Roulet (2005), Annual and seasonal variability in evapotranspiration and water table at a shrub-covered bog in southern Ontario, Canada, *Hydrol. Process.*, 19(18), 3533–3550, doi:10.1002/hyp.5842.
- Mackay, J. R. (1995), Active Layer Changes (1968 to 1993) Following the Forest-Tundra Fire near Inuvik, N. W. T., Canada, Arct. Alp. Res., 27(4), 323–336.

- Nossov, D. R., M. Torre Jorgenson, K. Kielland, and M. Z. Kanevskiy (2013), Edaphic and microclimatic controls over permafrost response to fire in interior Alaska, *Environ. Res. Lett.*, 8, 35013, doi:10.1088/1748-9326/8/3/035013.
- Olefeldt, D. et al. (2016), Circumpolar distribution and carbon storage of thermokarst landscapes, *Nat. Commun.*, 7, 13043, doi:10.1038/ncomms13043.
- Randerson, J. T. et al. (2006), The impact of boreal forest fire on climate warming., *Science*, *314*(5802), 1130–1132, doi:10.1126/science.1132075.
- Rocha, A. V, M. M. Loranty, P. E. Higuera, M. C. Mack, F. S. Hu, B. M. Jones, A. L. Breen, E.
  B. Rastetter, S. J. Goetz, and G. R. Shaver (2012), The footprint of Alaskan tundra fires during the past half-century: implications for surface properties and radiative forcing, *Environ. Res. Lett.*, 7(4), 44039, doi:10.1088/1748-9326/7/4/044039.
- Rowe, J., and G. W. Scotter (1973), Fire in the boreal forest, *Quat. Res.*, *3*, 444–464, doi:10.1016/0033-5894(73)90008-2.
- Schädel, C. et al. (2016), Potential carbon emissions dominated by carbon dioxide from thawed permafrost soils, *Nat. Clim. Chang.*, 6(October), doi:10.1038/NCLIMATE3054.
- Schaefer, K., H. Lantuit, V. E. Romanovsky, E. A. G. Schuur, and R. Witt (2012), The impact of the permafrost carbon feedback on gloval climate, *Environ. Res. Lett.*, 9, 1–9, doi:10.1088/1748-9326/9/8/085003.
- Schuur, E. A. G. et al. (2015), Climate change and the permafrost carbon feedback, *Nature*, *250*, 217–227, doi:10.1038/nature14338.
- Spiess, A., and N. Neumeyer (2010), An evaluation of R 2 as an inadequate measure for nonlinear models in pharmacological and biochemical research : a Monte Carlo approach, *BMC Pharmacol.*, 10(6), 1–11.

- Startsev, N., J. S. Bhatti, and R. S. Jassal (2016), Surface CO 2 Exchange Dynamics across a Climatic Gradient in McKenzie Valley: Effect of Landforms, Climate and Permafrost, *Forests*, 7(279), doi:10.3390/f7110279.
- Treat, C. C., D. Wisser, S. Marchenko, and S. Frolking (2013), Modelling the effects of climate change and disturbance on permafrost stability in northern organic soils, *Mires Peat*, 12, 1– 17.
- Turetsky, M. R., B. D. Amiro, E. Bosch, and J. S. Bhatti (2004), Historical burn area in western Canadian peatlands and its relationship to fire weather indices, *Global Biogeochem. Cycles*, 18(4), 1–9, doi:10.1029/2004GB002222.
- Turetsky, M. R., E. S. Kane, J. W. Harden, R. D. Ottmar, K. L. Manies, E. Hoy, and E. S. Kasischke (2011), Recent acceleration of biomass burning and carbon losses in Alaskan forests and peatlands, *Nat. Geosci.*, 4(1), 27–31, doi:10.1038/ngeo1027.
- Viereck, L. A., N. R. Werdin-pfisterer, and P. C. Adams (2008), Effect of Wildfire and Fireline Construction on the Annual Depth of Thaw in a Black Spruce Permafrost Forest in Interior Alaska : A 36-Year Record of Recovery, *Ninth Int. Conf. Permafr.*, 1845–1850.
- Wotton, B. M., C. a. Nock, and M. D. Flannigan (2010), Forest fire occurrence and climate change in Canada, *Int. J. Wildl. Fire*, *19*, 253–271, doi:10.1071/WF09002.
- Wright, N., M. Hayashi, and W. L. Quinton (2009), Spatial and temporal variations in active layer thawing and their implication on runoff generation in peat-covered permafrost terrain, *Water Resour. Res.*, 45(5), 1–13, doi:10.1029/2008WR006880.
- Xu, X., Y. Zhou, H. Ruan, Y. Luo, and J. Wang (2010), Temperature sensitivity increases with soil organic carbon recalcitrance along an elevational gradient in the Wuyi Mountains, China, *Soil Biol. Biochem.*, 42(10), 1811–1815, doi:10.1016/j.soilbio.2010.06.021.

- Yoshikawa, K., W. R. Bolton, V. E. Romanovsky, M. Fukuda, and L. D. Hinzman (2002), Impacts of wildfire on the permafrost in the boreal forests of Interior Alaska, J. Geophys. Res., 108(D1), 16–17, doi:10.1029/2001JD000438.
- Zhang, Y., S. A. Wolfe, P. D. Morse, I. Olthof, and R. H. Fraser (2015), Journal of Geophysical Research: Earth Surface, J. Geophys. Res. Earth Surf., (120), 2338–2356, doi:10.1002/2015JF003679.Received.



**Figure 3.1**: Peatland complexes and field sited in the Taiga Plains ecozone (inset) in western boreal Canada. Peatland plateaus within the Taiga Plains are a mosaic of treed permafrost peatlands and open, permafrost free thermokarst bogs. Study sites included two burned permafrost peat plateaus that burned during the summer of 2000 and 2007, and an unburn site



**Figure S3.1**: Soil moisture using a handheld soil moisture sensor and soil respiration. No relationship was found between soil respiration and moisture therefore soil moisture was not included in our models of soil temperature as a controlling factor.



**Figure S3.2**: Soil temperature, in two hour intervals, from hobo temperature loggers from shallow, 10cm (upper lines), and deep, 40cm (lower lines), during growing season. Temperature is averaged from the two loggers located within each site.



**Figure 3.2**: Soil temperature during growing season (JD 150-250). A) Soil temperatures at 10cm were significantly warmer in the burned than in the unburned (t=2.705, p<0.05). B) Soil temperatures at 40cm were also significantly higher in the burned than in the unburned (t=5.414,p<0.05) during the 2016 growing season.



**Figure 3.3**: Raw soil respiration during the growing season (JD 148-300). A) Soil respiration during the season with a second order polynomic trend line. Unburned sites displayed higher respiration during the early summer while burned sites showed increased respiration late summer. B) Exponential relationship between soil temperature and soil respiration.

Model	R2	AIC
$\mathbf{R} = [(\mathbf{a} * \mathbf{b}^{(T10/10)} + \mathbf{c} * \mathbf{b}^{(T40/10)}) * \mathbf{BURN}] + [(\mathbf{d} * \mathbf{b}^{(T10/10)} + \mathbf{e} * \mathbf{b}^{(T40/10)}) * \mathbf{UNBURN}]$	0.90	-34.21
$R = (a * b^{(T10/10)}*BURN) + (c * b^{(T10/10)}*UNBURN)$	0.88	-33.61
$R = [(a * Q_{10}^{(T5/10)} + c * b^{(T40/10)})*BURN] + [(d * b^{(T5/10)} + e * b^{(T40/10)})*UNBURN]$	0.88	-28.19
$R = (a * b^{(T5/10)}*BURN) + (c* b^{(T5/10)}*UNBURN)$	0.79	-16.40
$R = (a * b^{(T40/10)}*BURN) + (c * b^{(T40/10)}*UNBURN)$	0.70	-5.65

**Table 3.1**: Modeled equations for burned and unburned sites, arranged in order of decreasing model fit based on Akaike information criterion.

Table 3.2: Coefficients (±1SE) of variables in the best model based on an AIC evaluation.

Variable	Coefficient
a - Respiration at 0°C (10cm burned)	$0.14 \pm 0.05 \ \mu mol \ C \ m^{-2} \ s^{-1}$
$b - Q_{10}$	$2.56 \pm 0.31$
c – Respiration at 0°C (40cm burned)	$0.14 \pm 0.06 \ \mu mol \ C \ m^{-2} \ s^{-1}$
d – Respiration at 0°C (10cm unburned)	$0.30 \pm 0.08 \ \mu mol \ C \ m^{-2} \ s^{-1}$
e – Respiration at 0°C (40cm unburned)	$0.05 \pm 0.09 \ \mu mol \ C \ m^{-2} \ s^{-1}$



**Figure 3.4**: Modeled and measured soil respiration during the growing season (JD 140-300) in the burned (a) and unburned (b) sites. Model accurately predicted fluxes in the burned ( $R^2=0.79$ ) and the unburned ( $R^2=0.71$ ). Model performed most poorly at low temperatures and low fluxes, likely due to low n size at these conditions.



Figure 3.5: Fractional contribution of deep respiration to total soil respiration from the burned and unburned sites. Burned sites displayed increased respiration from depth later into the summer season whereas the unburned site remained relatively consistent, with only a minor increase.



**Figure 3.6**: A) Cumulative carbon loss (gC m<sup>2</sup>) from shallow soils (10cm) during the growing season (JD 150-258) with upper and lower bounds based on a  $Q_{10}$  of 2.56 ± 0.31. B) Cumulative carbon loss from deep soils (40cm) during the same period.

## **Chapter 4: Conclusion**

The goal of this thesis was to investigate how wildfire impacts permafrost stability and carbon cycling in northern permafrost peatlands. This work was motivated by projected changes to the fire regime [*Amiro et al.*, 2009] and permafrost stability [*Viereck et al.*, 2008; *Baltzer et al.*, 2014; *Mamet et al.*, 2017] within the discontinuous permafrost zone. Furthermore, the vast majority of studies that have begun to explore this issue have concentrated in Alaska peatlands, where peat properties and depths differ, thus making the extrapolation of their results to the circumpolar boreal region difficult. Therefore, this research focused on the discontinuous permafrost zone of the Taiga Plains in northwestern Canada where permafrost is found principally within peatlands where permafrost underlies peat plateaus that rise 0.5-1 m above surrounding permafrost free wetlands. These ecosystems also contain some of the most carbon rich soils within the circumpolar north [*Hugelius et al.*, 2013], thus making their response to disturbance very important for future climate conditions.

The first objective of this research was to understand over what temporal scale wildfire affects active layer depths. I found that wildfire's effect on active layer depths was twofold. On the short term, 0-10 years post fire, it causes a continuous deepening of the active layer. Follow this time period, as vegetation begins to recovers, the rate of thaw begins to decrease and active layers recover. However, a longer term effect of fire was also the formation of taliks, which were prominent from 10-20 years post fire. Therefore the net effect of fire can be seen for  $\sim$ 30-40 years at which time active layer depths have returned to pre-fire conditions and the proportion of taliks within the site is similar to that a unburned peat plateaus. This has important implications for the stability of the permafrost core and its net energy balance. We propose that due to the presence of these taliks, sufficient energy isn't able to escape during the winter which causes thawing at the edge of the permafrost core (objective 2).

The second object was to understand the extent that wildfire accelerates thermokarst development. I found that perturbed soil thermal conditions due to active layer deepening are likely responsible for a near tripling in recent thermokarst area within burned areas. I found that despite wildfire only affecting ~25% of permafrost peatlands within the study area, it is responsible for 25% of all thaw or 2,700km<sup>2</sup> of thawed permafrost thaw. Consequently, we conclude that it is important to incorporate the future role of wildfire in permafrost stability in

models for circumpolar boreal regions. Also, this thaw has important implications for traditional land-use, regional hydrology and water quality, and soil greenhouse gas emissions (objective 3).

The third object was to understand how soil respiration is affected following wildfire. I found that as a result of wildfire removing the top 5-20 cm of organic soils during a fire event [*Turetsky et al.*, 2011] removes the labile carbon pool burn sites do not become greater soils of carbon to the atmosphere. This is despite warmer soil temperatures at depth and increased active layers (ie. larger thawed soil carbon pools). However, by being one of the first to use a two temperature soil respiration model that incorporated a shallow and deep temperature we were able to partition the respiration and understand the magnitude of shallow and deep respiration (objective 3).

My final objective was to understand the relative proportion of respiration from depth in a burned site compared to an unburned site as this represents the respiration of deep, old, previously frozen carbon stores. I found that due to higher temperatures at depth deep respiration represented over 50% of all respiration within a burned site compared to just 14% of all respiration, representing over a fourfold increase in respiration from depth as a result of wildfire. This culminates in ~31.87  $\pm$  13.45 kg C m<sup>-2</sup> loss during the growing season (JD 150-258) from a burned peat plateau and only 8.60  $\pm$  12.13 kg C m<sup>-2</sup> from an unburned peat plateau. This highlights an importation potential phenomenon, that if active layers and soil temperatures increase with climate warming, in the absence of fire, an overall increased respiration could occur due to a potential 4 fold increase in respiration from depth.

Taken collectively, these conclusions show that wildfire can act as a major driver of change for both permafrost stability and carbon cycling within the discontinuous permafrost zone. Globally, permafrost peatlands cover 8% of the circumpolar world and store ~13Pg of soil carbon. By understanding their response to wildfire, we can better predict future permafrost conditions and carbon losses to the atmosphere. This will help us to make informed management and development decisions in the future.

## **Literature Cited**

- Amiro, B. D., Cantin, a., Flannigan, M. D., & de Groot, W. J. (2009). Future emissions from Canadian boreal forest fires. *Canadian Journal of Forest Research*, 39(2), 383–395. http://doi.org/10.1139/X08-154
- Baltzer, J. L., Veness, T., Chasmer, L. E., Sniderhan, A. E., & Quinton, W. L. (2014). Forests on thawing permafrost: Fragmentation, edge effects, and net forest loss. *Global Change Biology*, 20(3), 824–834. http://doi.org/10.1111/gcb.12349
- Hugelius, G., Tarnocai, C., Broll, G., Canadell, J. G., Kuhry, P., & Swanson, D. K. (2013). The Northern Circumpolar Soil Carbon Database: spatially distributed datasets of soil coverage and soil carbon storage in the northern permafrost regions. *Earth System Science Data*, 5, 3–13. http://doi.org/10.5194/essd-5-3-2013
- Mamet, S. D., Chun, K. P., Kershaw, G. G. L., Loranty, M. M., & Kerhaw, G. P. (2017). Recent Increases in Permafrost Thaw Rates and Areal Loss of Palsas in the Western Northwest Territories, Canada. *Permafrost and Periglacial Processes*. http://doi.org/10.1002/ppp.1951
- Turetsky, M. R., Kane, E. S., Harden, J. W., Ottmar, R. D., Manies, K. L., Hoy, E., & Kasischke,
  E. S. (2011). Recent acceleration of biomass burning and carbon losses in Alaskan forests and peatlands. *Nature Geoscience*, 4(1), 27–31. http://doi.org/10.1038/ngeo1027
- Viereck, L. A., Werdin-pfisterer, N. R., & Adams, P. C. (2008). Effect of Wildfire and Fireline Construction on the Annual Depth of Thaw in a Black Spruce Permafrost Forest in Interior Alaska : A 36-Year Record of Recovery. *Ninth International Conference on Permafrost*, 1845–1850.

#### **Bibliography**

- Amiro, B. D., A. Cantin, M. D. Flannigan, and W. J. de Groot (2009), Future emissions from Canadian boreal forest fires, *Can. J. For. Res.*, 39(2), 383–395, doi:10.1139/X08-154.
- Andrews, T. D., S. V Kokelj, G. MacKay, J. Buysse, I. Kritsch, A. Andre, and T. Lantz (2016), Permafrost Thaw and Aboriginal Cultural Landscapes in the Gwich'in Region, Canada, *APT Bull. J. Preserv. Technol.*, 47(1), 15–22.
- Baltzer, J. L., T. Veness, L. E. Chasmer, A. E. Sniderhan, and W. L. Quinton (2014), Forests on thawing permafrost: Fragmentation, edge effects, and net forest loss, *Glob. Chang. Biol.*, 20(3), 824–834, doi:10.1111/gcb.12349.
- Bates, D. M. and Chambers, J. M. (1992). Nonlinear models. In Chambers, J. M. and Hastie, T. J., editors, Statistical Models in S, pages 421–454. Wadsworth, Pacific Grove, CA.
- Beilman, D., and S. Robinson (2003), Peatland permafrost thaw and landform type along a climatic gradient, *Proc. 8th Int. Conf. Permafr.*, 61–65.
- Benscoter, B. W., and R. K. Wieder (2003), Variability in organic matter lost by combustion in a boreal bog during the 2001 Chisholm fire, *Can. J. For. Res.*, 33(12), 2509–2513, doi:10.1139/x03-162.
- Bergeron, Y. (2000), Species and Stand Dynamics in the Mixed Woods of Quebec Southern Boreal Forest, *Ecology*, *81*(6), 1500–1516
- Bridgham, S. D., K. Updegraff, and J. Pastor (1998), Carbon, nitrogen, and phosphorus mineralization in northern wetlands, *Ecology*, 79(5), 1545–1561, doi:10.1890/0012-9658(1998)079[1545:CNAPMI]2.0.CO;2.
- Bosatta, E., and G. I. Ågren (1999), Soil organic matter quality interpreted thermodynamically, *Soil Biol. Biochem.*, *31*(13), 1889–1891, doi:10.1016/S0038-0717(99)00105-4.
- Brown, D. R. N., M. T. Jorgenson, T. A. Douglas, V. E. Romanovsky, K. Kielland, C. Hiemstra,
  E. S. Euskirchen, and R. W. Ruess (2015), Interactive effects of wildfire and climate on
  permafrost degradation in Alaskan lowland forests, J. Geophys. Res., 120, 1619–1637,

doi:10.1002/2015JG003033.Received.

- Brown, J., O.J. Ferrians, Jr., J.A. Heginbottom, and E.S. Melnikov, eds. 1997. Circum-Arctic map of permafrost and ground-ice conditions. Washington, DC: U.S. Geological Survey in Cooperation with the Circum-Pacific Council for Energy and Mineral Resources. Circum-Pacific Map Series CP-45, scale 1:10,000,000, 1 sheet.
- Burn, C. R. (1998), The active layer: Two contrasting definitions, *Permafr. Periglac. Process.*, 9(4), 411–416.
- Callaghan, T.V., Bjorn, L.O., Chapin III, S.F.; Chernov, Y., Christensen, T.R., Huntley, B., Ims, R., Johansson, M., Riedlinger, D.J., Jonasson, S. Matveyeva, N., Oechel, W., Panikov, N., and Shaver, G. 2005. Arctic Tundra and Polar Desert Ecosystems. In Arctic Climate Impact Assessment, Cambridge University Press, Cambridge, United Kingdom.
- Canada, N. R. (2017), Canadian Wildland Fire Information System, Available from: http://cwfis.cfs.nrcan.gc.ca/datamart/metadata/nfdbpoly (Accessed 14 July 2017)
- Chapin, F. S., M. Sommerkorn, M. D. Robards, and K. Hillmer-Pegram (2015), Ecosystem stewardship: A resilience framework for arctic conservation, *Glob. Environ. Chang.*, 34, 207–217, doi:10.1016/j.gloenvcha.2015.07.003.
- Chasmer, L. E., and C. Hopkinson (2016), Threshold loss of discontinuous permafrost and landscape evolution, *Global Biogeochem. Cycles*, *doi:* 10.11, 1–15, doi:10.1111/gcb.13537.
- Chivers, M. R., M. R. Turetsky, J. M. Waddington, J. W. Harden, and a. D. McGuire (2009), Effects of experimental water table and temperature manipulations on ecosystem CO2 fluxes in an Alaskan rich fen, *Ecosystems*, 12(8), 1329–1342, doi:10.1007/s10021-009-9292-y.
- Christensen, T. R., S. Jonasson, T. V. Callaghan, and M. Havström (1999), On the potential CO2 release from tundra soils in a changing climate, *Appl. Soil Ecol.*, *11*(2–3), 127–134, doi:10.1016/S0929-1393(98)00146-2.
- Conant, R. T., R. A. Drijber, M. L. Haddix, W. J. Parton, E. A. Paul, A. F. Plante, J. Six, and M.

J. Steinweg (2008), Sensitivity of organic matter decomposition to warming varies with its quality, *Glob. Chang. Biol.*, *14*(4), 868–877, doi:10.1111/j.1365-2486.2008.01541.x.

- Cumming, S. G. (2001), Forest Type and Wildfire in the Alberta Boreal Mixedwood : What Do Fires Burn, *Ecol. Appl.*, *11*(1), 97–110.
- Davidson, E. A., and I. A. Janssens (2006), Temperature sensitivity of soil carbon decomposition and feedbacks to climate change, *Nature*, *440*(7081), 165–173, doi:10.1038/nature04514.
- Dorrepaal, E., S. Toet, R. S. P. van Logtestijn, E. Swart, M. J. van de Weg, T. V. Callaghan, and R. Aerts (2009), Carbon respiration from subsurface peat accelerated by climate warming in the subarctic, *Nature*, 460(7255), 616–619, doi:10.1038/nature08216.
- Dutta, K., E. A. G. Schuur, J. C. Neff, and S. A. Zimov (2006), Potential carbon release from permafrost soils of Northeastern Siberia, *Glob. Chang. Biol.*, 12(12), 2336–2351, doi:10.1111/j.1365-2486.2006.01259.x.
- Flannigan, M., B. Stocks, M. Turetsky, and M. Wotton (2009), Impacts of climate change on fire activity and fire management in the circumboreal forest, *Glob. Chang. Biol.*, 15(3), 549– 560, doi:10.1111/j.1365-2486.2008.01660.x.
- Flannigan, M., A. S. Cantin, W. J. De Groot, M. Wotton, A. Newbery, and L. M. Gowman (2013), Global wildland fire season severity in the 21st century, *For. Ecol. Manage.*, 294, 54–61, doi:10.1016/j.foreco.2012.10.022.
- Frey, K. E., and J. w. McClelland (2009), Impacts of permafrost degradation on arctic river biogeochemistry, *Hydrol. Process.*, 23, 169–182, doi:10.1002/hyp.
- Genet, H. et al. (2013), Modeling the effects of fire severity and climate warming on active layer thickness and soil carbon storage of black spruce forests across the landscape in interior Alaska, *Environ. Res. Lett.*, *8*, 1–13, doi:10.1088/1748-9326/8/4/045016.

Gignac, L. D., D. H. Vitt, S. C. Zoltai, and S. E. Bayley (1991), Bryophyte Response Surfaces

Along Climatic, Chemical, and Physical Gradients in Peatlands of Western Canada, *Nov. Hedwigia*, *53*(1–2), 27–71.

- Gibson, C. M., M. R. Turetsky, K. Cottenie, E. S. Kane, G. Houle, and E. S. Kasischke (2016), Variation in plant community composition and vegetation carbon pools a decade following a severe fire season in interior Alaska, *J. Veg. Sci.*, 1–11, doi:10.1111/jvs.12443.
- Gleason, K. E., A. W. Nolin, and T. R. Roth (2013), Charred forests increase snowmelt : Effects of burned woody debris and incoming solar radiation on snow ablation, *Geophys. Res. Lett.*, 40, 4654–4661, doi:10.1002/grl.50896.
- Group, E. S. W. (1995), A National Ecological Framework for Canada.
- Goulden, M. L. (1998), Sensitivity of Boreal Forest Carbon Balance to Soil Thaw, *Science (80-. ).*, *279*(5348), 214–217, doi:10.1126/science.279.5348.214.
- Gorham, E. (1991), Northern Peatlands: Role in the Carbon Cycle and Probable Responses To Climatic Warming, *Ecol. Appl.*, *1*(2), 182–195.
- Hedstrom, N. R., and J. W. Pomeroy (1998), Measurements and modeling of snow interception in the boreal forest, *Hydrol. Process.*, *12*, 1611–1625.
- Heginbottom, J.A., and M.A. Dubreuil. 1995. Canada -- Permafrost in *National Atlas of Canada*, 5th edition, Plate 2.1 (MCR 4177), scale 1:7,500,000.
- Helbig, M., K. Wischnewski, N. Kljun, and E. Laura (2016), Regional atmospheric cooling and wetting effect of permafrost thaw-induced boreal forest loss, *Glob. Chang. Biol.*, 22, 4048– 4066, doi:10.1111/gcb.13348.
- Hilasvuori, E., A. Akujärvi, H. Fritze, K. Karhu, R. Laiho, P. Mäkiranta, M. Oinonen, V. Palonen, P. Vanhala, and J. Liski (2013), Temperature sensitivity of decomposition in a peat profile, *Soil Biol. Biochem.*, 67, 47–54, doi:10.1016/j.soilbio.2013.08.009
- Hovelsrud, G. K., B. Poppel, B. Van Oort, and J. D. Reist (2011), Arctic societies, cultures, and peoples in a changing cryosphere, *Ambio*, 40, 100–110, doi:10.1007/s13280-011-0219-4.

- Hugelius, G., C. Tarnocai, G. Broll, J. G. Canadell, P. Kuhry, and D. K. Swanson (2013), The Northern Circumpolar Soil Carbon Database: spatially distributed datasets of soil coverage and soil carbon storage in the northern permafrost regions, *Earth Syst. Sci. Data*, 5, 3–13, doi:10.5194/essd-5-3-2013.
- Hugelius, G. et al. (2014), Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps, *Biogeosciences*, *11*(23), 6573–6593, doi:10.5194/bg-11-6573-2014.
- Islam, M. A., and S. E. Macdonald (2004), Ecophysiological adaptations of black spruce (Picea mariana) and tamarack (Larix laricina) seedlings to flooding, *Trees Struct. Funct.*, 18(1), 35–42, doi:10.1007/s00468-003-0276-9.
- IPCC, 20016: Climate Change 20016: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M.Tignor and H.L. Miller (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Jafarov, E., and V. Romanovsky (2013), The effects of fire on the thermal stability of permafrost in lowland and upland black spruce forests of interior Alaska in a changing climate, *Environ. Res. Lett.*, 8(3), doi:10.1088/1748-9326/8/3/035030.
- Johnstone, J. F., and F. S. Chapin (2006), Effects of soil burn severity on post-fire tree recruitment in boreal forest, *Ecosystems*, 9(1), 14–31, doi:10.1007/s10021-004-0042-x.
- Johnstone, J. F., T. N. Hollingsworth, F. S. Chapin, and M. C. Mack (2010), Changes in fire regime break the legacy lock on successional trajectories in Alaskan boreal forest, *Glob. Chang. Biol.*, 16(4), 1281–1295, doi:10.1111/j.1365-2486.2009.02051.x.
- Jonasson, G. (1983), The point intercept method for non-destructive estimation of biomass, *Phytocoenologia*, *11*(3), 385–388, doi:10.2307/3565988.
- Kasischke, E. S., and M. R. Turetsky (2006), Recent changes in the fire regime across the North American boreal region—Spatial and temporal patterns of burning across Canada and

Alaska, Geophys. Res. Lett., 33(9), L09703, doi:10.1029/2006GL025677.

- Kettridge, N., D. K. Thompson, and J. M. Waddington (2012), Impact of wildfire on the thermal behavior of northern peatlands: Observations and model simulations, J. Geophys. Res. Biogeosciences, 117, doi:10.1029/2011JG001910.
- Koven, C. D., D. M. Lawrence, and W. J. Riley (2015), Permafrost carbon climate feedback is sensitive to deep soil carbon decomposability but not deep soil nitrogen dynamics, *Proc. Natl. Acad. Sci.*, *112*(12), 3752–3757, doi:10.1073/pnas.1415123112.
- Kirschbaum, M. U. F. (1995), the Temperature Dependence of Soil Organic Matter Decomposition, and the Effect of Global Warming on Soil Organic C Storage, *Soil Biol. Biochem.*, 27(6), 753–760.
- Knorr, W., I. Prentice, J. House, and E. Holland (2005), Long-term sensitivity of soil carbon turnover to warming, *Nature*, 433(January), 298–301, doi:10.129/2002PA000837.
- Kuhry, P. (1994), of Sphagnum- in the development The role of fire boreal Canada in western peatlands dominated, *J. Ecol.*, *82*(4), 899–910.
- Lafleur, P. M., R. a. Hember, S. W. Admiral, and N. T. Roulet (2005), Annual and seasonal variability in evapotranspiration and water table at a shrub-covered bog in southern Ontario, Canada, *Hydrol. Process.*, 19(18), 3533–3550, doi:10.1002/hyp.5842.
- Lawrence, D. M., A. G. Slater, and S. C. Swenson (2012), Simulation of present-day and future permafrost and seasonally frozen ground conditions in CCSM4, *J. Clim.*, 25(7), 2207–2225, doi:10.1175/JCLI-D-11-00334.1.
- Lee, H., E. A. G. Schuur, K. S. Inglett, M. La Voie, and J. P. Chanton (2012), The rate of permafrost carbon release under aerobic and anaerobic conditions and its potential effects on climate, *Glob. Chang. Biol.*, 18, 515–527, doi:10.1111/j.1365-2486.2011.02519.x.
- Loisel, J. et al. (2014), A database and synthesis of northern peatland soil properties and Holocene carbon and nitrogen accumulation, *The Holocene*, 24(9), 1028–1042,

doi:10.1177/0959683614538073.

- Mackay, J. R. (1995), Active Layer Changes (1968 to 1993) Following the Forest-Tundra Fire near Inuvik, N. W. T., Canada, Arct. Alp. Res., 27(4), 323–336.
- Mamet, S. D., K. P. Chun, G. G. L. Kershaw, M. M. Loranty, and G. P. Kerhaw (2017), Recent Increases in Permafrost Thaw Rates and Areal Loss of Palsas in the Western Northwest Territories, Canada, *Permafr. Periglac. Process.*, doi:10.1002/ppp.1951.
- McKenzie, C., S. Schiff, R. Aravena, C. Kelly, and V. St. Louis (1998), Effect of temperature on production of CH4 and CO2 from peat in a natural and flooded boreal forest wetland, *Clim. Change*, 40, 247–266, doi:10.1023/A:1005416903368.
- Melvin, A. M. et al. (2016), Climate change damages to Alaska public infrastructure and the economics of proactive adaptation, *Proc. Natl. Acad. Sci.*, 114(2), 122–131, doi:10.1073/pnas.1611056113.
- Moore, T. R., and M. Dalva (1993), The influence of temperature and water table position on carbon dioxide and methane emissions from laboratory columns of peatland soils, *J. Soil Sci.*, 44(4), 651–664, doi:10.1111/j.1365-2389.1993.tb02330.x.
- Nossov, D. R., M. T. Jorgenson, K. Kielland, K. D. Johnson, J. W. Harden, A. D. Mcguire, G. Grosse, S. Goetz, and A. D. Mcguire (2013), The effects of fire on the thermal stability of permafrost in lowland and upland black spruce forests of interior Alaska in a changing climate, *Environ. Res. Lett.*, (8), 1–11, doi:10.1088/1748-9326/8/3/035030.
- Olefeldt, D., and N. T. Roulet (2014), Permafrost conditions in peatlands regulate magnitude, timing, and chemical composition of catchment dissolved organic carbon export, *Glob. Chang. Biol.*, *20*(10), 3122–3136, doi:10.1111/gcb.12607.
- Olefeldt, D., M. R. Turetsky, P. M. Crill, and a. D. Mcguire (2013), Environmental and physical controls on northern terrestrial methane emissions across permafrost zones, *Glob. Chang. Biol.*, 19(2), 589–603, doi:10.1111/gcb.12071.
- Olefeldt, D. et al. (2016), Circumpolar distribution and carbon storage of thermokarst landscapes, *Nat. Commun.*, 7, 13043, doi:10.1038/ncomms13043.

- O'Neill, K. P., E. S. Kasischke, and D. D. Richter (2002), flux following fire in black spruce, white spruce, and aspen stands of interior Alaska, *Can. J. For. Res.*, *32*(9), 1525–1541, doi:10.1139/x02-077.
- Payette, S., H. Samson, and D. Lagarec (1976), The evolution of permafrost in the taiga and in the forest-tundra, wester Quebec-Labrador Peninsula, *Can. J. For. Res.*, 6(2), 203–220, doi:10.1007/s13398-014-0173-7.2.
- Pelletier, N., J. Talbot, D. Olefeldt, M. R. Turetsky, C. Blodau, O. Sonnentag, and W. L. Quinton (2017), Influence of Holocene permafrost aggradation and thaw on the paleoecology and carbon storage of a peatland complex in northwestern Canada, *The Holocene*, 1–15, doi:10.1177/0959683617693899.
- Quinton, W. L., M. Hayashi, and L. E. Chasmer (2009), Peatland Hydrology of Discontinuous Permafrost in the Northwest Territories: Overview and Synthesis, *Can. Water Resour. J.*, 34(4), 311–328, doi:10.4296/cwrj3404311.
- Quinton, W. L., M. Hayashi, and L. E. Chasmer (2011), Permafrost-thaw-induced land-cover change in the Canadian subarctic: implications for water resources, *Hydrol. Process.*, 25(1), 152–158, doi:10.1002/hyp.7894.
- Randerson, J. T. et al. (2006), The Impact of Boreal Forest Fire on Climate Warming, *Science* (80-.)., 314(5802), 1130–1132, doi:10.1126/science.1132075.
- Rocha, A. V, M. M. Loranty, P. E. Higuera, M. C. Mack, F. S. Hu, B. M. Jones, A. L. Breen, E.
  B. Rastetter, S. J. Goetz, and G. R. Shaver (2012), The footprint of Alaskan tundra fires during the past half-century: implications for surface properties and radiative forcing, *Environ. Res. Lett.*, 7(4), 44039, doi:10.1088/1748-9326/7/4/044039.
- Rowe, J., and G. W. Scotter (1973), Fire in the boreal forest, *Quat. Res.*, *3*, 444–464, doi:10.1016/0033-5894(73)90008-2.
- Seppälä, M. 1988. Palsas and related forms. In: Clark, M.J. (ed.). Advances in Periglacial Geomorphology, pp. 247–278. John Wiley and Sons, Chichester
- Savo, V., D. Lepofsky, H. Bay, and H. Bay (2017), Impacts of Climate Change on Subsistence-

Oriented Communities, Elsevier Inc.

- Schädel, C. et al. (2016), Potential carbon emissions dominated by carbon dioxide from thawed permafrost soils, *Nat. Clim. Chang.*, 6(October), doi:10.1038/NCLIMATE3054.
- Schaefer, K., H. Lantuit, V. E. Romanovsky, E. A. G. Schuur, and R. Witt (2012), The impact of the permafrost carbon feedback on gloval climate, *Environ. Res. Lett.*, 9, 1–9, doi:10.1088/1748-9326/9/8/085003
- Schuur, E., and J. Bockheim (2008), Vulnerability of permafrost carbon to climate change: Implications for the global carbon cycle, *Bioscience*, 58(September), 701–714, doi:10.1641/B580807.
- Schuur, E. A. G. et al. (2015), Climate change and the permafrost carbon feedback, *Nature*, *250*, 217–227, doi:10.1038/nature14338.
- Smith, S. L., D. W. Riseborough, and P. P. Bonnaventure (2015), Eighteen Year Record of Forest Fire Effects on Ground Thermal Regimes and Permafrost in the Central Mackenzie Valley, NWT, Canada, *Permafr. Periglac. Process.*, 26, 289–303, doi:10.1002/ppp.1849.
- Spiess, A., and N. Neumeyer (2010), An evaluation of R 2 as an inadequate measure for nonlinear models in pharmacological and biochemical research : a Monte Carlo approach, *BMC Pharmacol.*, 10(6), 1–11.
- Startsev, N., J. S. Bhatti, and R. S. Jassal (2016), Surface CO 2 Exchange Dynamics across a Climatic Gradient in McKenzie Valley: Effect of Landforms, Climate and Permafrost, *Forests*, 7(279), doi:10.3390/f7110279.
- Stocks, B. J. et al. (2002), Large forest fires in Canada, 1959–1997, J. Geophys. Res., 108, doi:10.1029/2001JD000484.
- Tarnocai, C., F. Mark Nixon, and L. Kutny (2004), Circumpolar-Active-Layer-Monitoring(CALM) sites in the Mackenzie Valley, northwestern Canada, *Permafr. Periglac. Process.*, 15(2), 141–153, doi:10.1002/ppp.490.

- Thie, J. (1974), Distribution and thawing of permafrost in the southern part of the discontinuous permafrost zone in Manitoba, *Arctic*, *27*(3), 189–200.
- Thompson, D. K., A. S. Baisley, and J. M. Waddington (2015), Seasonal variation in albedo and radiation exchange between a burned and unburned forested peatland: implications for peatland evaporation, *Hydrol. Process.*, 29(14), 3227–3235, doi:10.1002/hyp.10436.
- Tolonen, K., and J. Turunen (1996), Accumulation rates of carbon in mires in Finland and implications for climate change, *The Holocene*, *6*(2), 171–178, doi:10.1177/095968369600600204.
- Treat, C. C., D. Wisser, S. Marchenko, and S. Frolking (2013), Modelling the effects of climate change and disturbance on permafrost stability in northern organic soils, *Mires Peat*, 12, 1– 17.
- Turetsky, M. (2002), Current disturbance and the diminishing peatland carbon sink, *Geophys. Res. Lett.*, *29*(11), 7–10, doi:10.1029/2001GL014000.
- Turetsky, M. R., and R. K. Wieder (2001), A direct approach to quantifying organic matter lost as a result of peatland wildfire, *Can. J. For. Res.*, *31*(2), 363–366, doi:10.1139/x00-170.
- Turetsky, M. R., R. K. Wieder, and D. H. Vitt (2002), Boreal peatland C fluxes under varying permafrost regimes, *Soil Biol. Biochem.*, 34(7), 907–912, doi:10.1016/S0038-0717(02)00022-6.
- Turetsky, M. R., B. D. Amiro, E. Bosch, and J. S. Bhatti (2004), Historical burn area in western Canadian peatlands and its relationship to fire weather indices, *Global Biogeochem. Cycles*, 18(4), 1–9, doi:10.1029/2004GB002222.
- Turetsky, M. R., R. K. Wieder, D. H. Vitt, R. J. Evans, and K. D. Scott (2007), The disappearance of relict permafrost in boreal North America: Effects on peatland carbon storage and fluxes, *Glob. Chang. Biol.*, 13, 1922–1934, doi:10.1111/j.1365-2486.2007.01381.x.

Turetsky, M. R., E. S. Kane, J. W. Harden, R. D. Ottmar, K. L. Manies, E. Hoy, and E. S.

Kasischke (2011), Recent acceleration of biomass burning and carbon losses in Alaskan forests and peatlands, *Nat. Geosci.*, *4*(1), 27–31, doi:10.1038/ngeo1027.

- Viereck, L. A., N. R. Werdin-pfisterer, and P. C. Adams (2008), Effect of Wildfire and Fireline Construction on the Annual Depth of Thaw in a Black Spruce Permafrost Forest in Interior Alaska : A 36-Year Record of Recovery, *Ninth Int. Conf. Permafr.*, 1845–1850.
- Vitt, D. H., Y. Li, and R. J. Belland (1995), Patterns of bryophyte diversity in peatlands of continental western Canada, *Bryologist*, *98*(2), 218–227, doi:10.2307/3243306.
- Vitt, D. H., L. a Halsey, I. E. Bauer, and C. Campbell (2000), Spatial and temporal trends in carbon storage of peatlands of continental western Canada through the Holocene, *Can. J. Earth Sci.*, 37(5), 683–693, doi:10.1139/e99-097.
- Walsh, J.E., Anisimov, O.H., Hagen, J.O.M., Jakobsson T., Oerlemans, J., Prowse, T.D., Romanovsky, V., Savelieva, N., Serreze, M., Shiklomanov, A., Shiklomanov, I. and Solomon, S. 2005. Cryosphere and Hydrology, In Arctic Climate Impact Assessment, Cambridge University Press, Cambridge, United Kingdom.
- Walter, K. M., S. A. Zimov, J. P. Chanton, D. Verbyla, and F. S. Chapin (2006), Methane bubbling from Siberian thaw lakes as a positive feedback to climate warming, *Nature*, 443(7107), 71–75, doi:10.1038/nature05040.
- Woo, M., K. Young, and L. Brown (2006), High Arctic Patchy Wetlands: Hydrologic Variability and Their Sustainability, *Phys. Geogr.*, *27*(4), 297–307, doi:10.2747/0272-3646.27.4.297.
- Wotton, B. M., C. a. Nock, and M. D. Flannigan (2010), Forest fire occurrence and climate change in Canada, *Int. J. Wildl. Fire*, *19*, 253–271, doi:10.1071/WF09002.
- Wright, N., M. Hayashi, and W. L. Quinton (2009), Spatial and temporal variations in active layer thawing and their implication on runoff generation in peat-covered permafrost terrain, *Water Resour. Res.*, 45(5), 1–13, doi:10.1029/2008WR006880.
- Xu, X., Y. Zhou, H. Ruan, Y. Luo, and J. Wang (2010), Temperature sensitivity increases with soil organic carbon recalcitrance along an elevational gradient in the Wuyi Mountains, China, *Soil Biol. Biochem.*, 42(10), 1811–1815, doi:10.1016/j.soilbio.2010.06.021.
- Yavitt, J. B., C. J. Williams, and R. K. Wieder (1997), Production of methane and carbon dioxide in peatland ecosystems across North America: Effects of temperature, aeration, and organic chemistry of peat, *Geomicrobiol. J.*, *14*(September), 299–316, doi:10.1080/01490459709378054.
- Yoshikawa, K., W. R. Bolton, V. E. Romanovsky, M. Fukuda, and L. D. Hinzman (2002), Impacts of wildfire on the permafrost in the boreal forests of Interior Alaska, J. Geophys. Res., 108(D1), 16–17, doi:10.1029/2001JD000438.
- Zhang, T. (2005), Influence of the Seasonal Snow Cover on the Ground Thermal Regime : an Overview, *Rev. Geophys.*, *43*, 1–23, doi:10.1029/2004RG000157.1.INTRODUCTION.
- Zhang, Y., W. Chen, and D. W. Riseborough (2008), Transient projections of permafrost distribution in Canada during the 21st century under scenarios of climate change, *Glob. Planet. Change*, 60, 443–456, doi:10.1016/j.gloplacha.2007.05.003.
- Zhang, Y., S. A. Wolfe, P. D. Morse, I. Olthof, and R. H. Fraser (2015), Journal of Geophysical Research : Earth Surface, J. Geophys. Res. Earth Surf., (120), 2338–2356, doi:10.1002/2015JF003679.Received.
- Zoltai, S. C. (1972), Palsas and Peat Plateaus in Central Manitoba and Saskatchewan, *Can. J. For. Res.*, 2(3), 291–302, doi:10.1139/x72-046.
- Zoltai, S. C. (1993), Cyclic Development of Permafrost in the Peatlands of Northwestern Canada, Arct. Alp. Res., 25(3), 240–246, doi:1df0.2307/1551820.